IAEA Safety Standards for protecting people and the environment

Operational Limits and Conditions and Operating Procedures for Research Reactors

Safety Guide No. NS-G-4.4





OPERATIONAL LIMITS AND CONDITIONS AND OPERATING PROCEDURES FOR RESEARCH REACTORS

Safety standards survey

The IAEA welcomes your response. Please see: http://www-ns.iaea.org/standards/feedback.htm The following States are Members of the International Atomic Energy Agency:

AFGHANISTAN ALBANIA ALGERIA ANGOLA ARGENTINA ARMENIA AUSTRALIA AUSTRIA AZERBAHAN BANGLADESH BELARUS BELGIUM BELIZE BENIN BOLIVIA BOSNIA AND HERZEGOVINA BOTSWANA BRAZIL BULGARIA BURKINA FASO CAMEROON CANADA CENTRAL AFRICAN REPUBLIC CHAD CHILE CHINA COLOMBIA COSTA RICA CÔTE D'IVOIRE CROATIA CUBA CYPRUS CZECH REPUBLIC DEMOCRATIC REPUBLIC OF THE CONGO DENMARK DOMINICAN REPUBLIC ECUADOR EGYPT EL SALVADOR ERITREA **ESTONIA ETHIOPIA** FINLAND FRANCE GABON GEORGIA GERMANY GHANA

GREECE **GUATEMALA** HAITI HOLY SEE HONDURAS HUNGARY ICELAND INDIA INDONESIA IRAN. ISLAMIC REPUBLIC OF IR AO IRELAND ISRAEL ITALY JAMAICA JAPAN JORDAN KAZAKHSTAN KENYA KOREA, REPUBLIC OF KUWAIT KYRGYZSTAN LATVIA LEBANON LIBERIA LIBYAN ARAB JAMAHIRIYA LIECHTENSTEIN LITHUANIA LUXEMBOURG MADAGASCAR MALAWI MALAYSIA MALI MALTA MARSHALL ISLANDS MAURITANIA MAURITIUS MEXICO MONACO MONGOLIA MONTENEGRO MOROCCO MOZAMBIQUE MYANMAR NAMIBIA NETHERLANDS NEW ZEALAND NICARAGUA NIGER NIGERIA

NORWAY PAKISTAN PALAU PANAMA PARAGUAY PERU PHILIPPINES POLAND PORTUGAL OATAR REPUBLIC OF MOLDOVA ROMANIA RUSSIAN FEDERATION SAUDI ARABIA SENEGAL SERBIA SEYCHELLES SIERRA LEONE SINGAPORE **SLOVAKIA** SLOVENIA SOUTH AFRICA SPAIN SRI LANKA SUDAN SWEDEN SWITZERLAND SYRIAN ARAB REPUBLIC TAJIKISTAN THAILAND THE FORMER YUGOSLAV REPUBLIC OF MACEDONIA TUNISIA TURKEY UGANDA UKRAINE UNITED ARAB EMIRATES UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND UNITED REPUBLIC OF TANZANIA UNITED STATES OF AMERICA URUGUAY UZBEKISTAN VENEZUELA VIETNAM YEMEN ZAMBIA ZIMBABWE

The Agency's Statute was approved on 23 October 1956 by the Conference on the Statute of the IAEA held at United Nations Headquarters, New York; it entered into force on 29 July 1957. The Headquarters of the Agency are situated in Vienna. Its principal objective is "to accelerate and enlarge the contribution of atomic energy to peace, health and prosperity throughout the world".

IAEA SAFETY STANDARDS SERIES No. NS-G-4.4

OPERATIONAL LIMITS AND CONDITIONS AND OPERATING PROCEDURES FOR RESEARCH REACTORS

SAFETY GUIDE

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA, 2008

COPYRIGHT NOTICE

All IAEA scientific and technical publications are protected by the terms of the Universal Copyright Convention as adopted in 1952 (Berne) and as revised in 1972 (Paris). The copyright has since been extended by the World Intellectual Property Organization (Geneva) to include electronic and virtual intellectual property. Permission to use whole or parts of texts contained in IAEA publications in printed or electronic form must be obtained and is usually subject to royalty agreements. Proposals for non-commercial reproductions and translations are welcomed and considered on a case-by-case basis. Enquiries should be addressed to the IAEA Publishing Section at:

Sales and Promotion, Publishing Section International Atomic Energy Agency Wagramer Strasse 5 P.O. Box 100 1400 Vienna, Austria fax: +43 1 2600 29302 tel.: +43 1 2600 22417 email: sales.publications@iaea.org http://www.iaea.org/books

> © IAEA, 2008 Printed by the IAEA in Austria

July 2008 STI/PUB/1333

IAEA Library Cataloguing in Publication Data

Operational limits and conditions and operating procedures for research reactors : safety guide. — Vienna : International Atomic Energy Agency, 2008. p. ; 24 cm. — (IAEA safety standards series, ISSN 1020-525X ; no. NS-G-4.4) STI/PUB/1333 ISBN 978–92–0–103508–0 Includes bibliographical references.

1. Nuclear reactors — Safety measures. I. International Atomic Energy Agency. II. Series.

IAEAL

08-00514

FOREWORD

by Mohamed ElBaradei Director General

The IAEA's Statute authorizes the Agency to establish safety standards to protect health and minimize danger to life and property — standards which the IAEA must use in its own operations, and which a State can apply by means of its regulatory provisions for nuclear and radiation safety. A comprehensive body of safety standards under regular review, together with the IAEA's assistance in their application, has become a key element in a global safety regime.

In the mid-1990s, a major overhaul of the IAEA's safety standards programme was initiated, with a revised oversight committee structure and a systematic approach to updating the entire corpus of standards. The new standards that have resulted are of a high calibre and reflect best practices in Member States. With the assistance of the Commission on Safety Standards, the IAEA is working to promote the global acceptance and use of its safety standards.

Safety standards are only effective, however, if they are properly applied in practice. The IAEA's safety services — which range in scope from engineering safety, operational safety, and radiation, transport and waste safety to regulatory matters and safety culture in organizations — assist Member States in applying the standards and appraise their effectiveness. These safety services enable valuable insights to be shared and I continue to urge all Member States to make use of them.

Regulating nuclear and radiation safety is a national responsibility, and many Member States have decided to adopt the IAEA's safety standards for use in their national regulations. For the Contracting Parties to the various international safety conventions, IAEA standards provide a consistent, reliable means of ensuring the effective fulfilment of obligations under the conventions. The standards are also applied by designers, manufacturers and operators around the world to enhance nuclear and radiation safety in power generation, medicine, industry, agriculture, research and education.

The IAEA takes seriously the enduring challenge for users and regulators everywhere: that of ensuring a high level of safety in the use of nuclear materials and radiation sources around the world. Their continuing utilization for the benefit of humankind must be managed in a safe manner, and the IAEA safety standards are designed to facilitate the achievement of that goal.

EDITORIAL NOTE

An appendix, when included, is considered to form an integral part of the standard and to have the same status as the main text. Annexes, footnotes and bibliographies, if included, are used to provide additional information or practical examples that might be helpful to the user.

The safety standards use the form 'shall' in making statements about requirements, responsibilities and obligations. Use of the form 'should' denotes recommendations of a desired option.

CONTENTS

1.	INTRODUCTION	1
	Background (1.1–1.5) Objective (1.6) Scope (1.7–1.13) Structure (1.14)	1 3 3 5
2.	DEVELOPMENT OF OPERATIONAL LIMITS AND CONDITIONS	6
	General (2.1–2.15) Attributes of operational limits and conditions (2.16–2.27)	6 10
3.	CONTENT OF THE OPERATIONAL LIMITS AND CONDITIONS DOCUMENT (3.1)	12
	Table of contents (3.2)Definitions (3.3)Introduction (3.4)Safety limits (3.5–3.13)Safety system settings (3.14–3.18)Limiting conditions for safe operation (3.19–3.26)Surveillance requirements (3.27–3.32)Administrative requirements (3.33–3.43)	13 13 13 13 15 16 18 19
4.	DEVELOPMENT OF OPERATING PROCEDURES	22
	General (4.1–4.5) Roles and responsibilities (4.6–4.20) Categorization and list of operating procedures (4.21–4.22) General considerations for the development of operating procedures (4.23–4.29)	22 23 25 26
	Steps in the preparation of a procedure (4.30–4.38)	28

5.	FORMAT AND CONTENT	
	OF OPERATING PROCEDURES	29
	General (5.1–5.9)	29
	Commissioning procedures (5.10–5.22)	32
	Operational procedures (5.23–5.27)	33
	Maintenance procedures (5.28–5.35)	34
	Inspection, calibration and periodic testing procedures (5.36–5.43)	35
	Radiation protection procedures (5.44–5.47)	36
	Procedures for the authorization of operation,	50
	maintenance and utilization (5.48–5.50)	37
	Procedures for operator response to anticipated	51
	operational occurrences (5.51–5.52)	38
	Emergency procedures (5.53–5.56)	38
	Physical protection procedures (5.57–5.60)	39
	Procedures for the handling of radioactive waste and	57
	control of radioactive releases (5.61–5.63)	40
	Procedures for extended shutdown (5.64–5.69)	40
	Procedures for utilization and for modification	10
	of the reactor (5.70–5.79)	41
	Administrative procedures (5.80–5.83)	43
6.	TRAINING OF PERSONNEL	
	IN THE USE OF PROCEDURES (6.1–6.5)	44
7.	COMPLIANCE WITH OPERATIONAL LIMITS	
	AND CONDITIONS AND OPERATING	
	PROCEDURES (7.1–7.4)	45
8.	QUALITY ASSURANCE	46
	General (8.1–8.4)	46
	Specific quality assurance arrangements for operational limits	
	and conditions (8.5–8.6)	47
	Quality assurance for operating procedures (8.7–8.8)	47
AP	PENDIX I: FACTORS TO BE CONSIDERED	
	IN ESTABLISHING LIMITING CONDITIONS	
	FOR SAFE OPERATION	49

APPENDIX II: INDICATIVE LIST OF OPERATING	
PROCEDURES	54
REFERENCES	59
CONTRIBUTORS TO DRAFTING AND REVIEW	61
BODIES FOR THE ENDORSEMENT OF	
IAEA SAFETY STANDARDS	65

1. INTRODUCTION

BACKGROUND

1.1. This Safety Guide was developed under the IAEA work programme for safety standards for research reactors. It supplements and elaborates upon the safety requirements for operational limits and conditions (OLCs) and operating procedures for research reactors that are established in Section 7 (paras 7.29–7.41, 7.51–7.55) of the IAEA publication on the Safety of Research Reactors [1].

1.2. For a research reactor to be operated safely, the provisions made in the final design and any subsequent modifications to the design must be taken into account in specifying the limits on operating parameters and the requirements on the equipment and personnel of the reactor facility. Under the responsibility of the operating organization, these limits on operating parameters and requirements should be developed in the evaluation of the design safety as a set of OLCs.¹ A major contribution to compliance with the OLCs is made by the development and utilization of operating procedures that are consistent with the OLCs. Establishment of OLCs and operating procedures is essential to achieving the objectives of the defence in depth concept.²

¹ The terms 'safety specifications' or 'technical specifications (tech specs) for safe operation' and 'general operating rules' are sometimes used to mean OLCs. These terms are used by operating organizations and regulatory bodies for nuclear reactors. These expressions usually cover safety limits, safety system settings, limiting conditions for safe operation, and surveillance requirements and administrative requirements. In addition, in some Member States the term 'operating rules' is used to mean the equivalent of safety limits, safety system settings and limiting conditions for safe operation, and this does not then include surveillance requirements and administrative requirements.

² Defence in depth is a hierarchical deployment of different levels of diverse equipment and procedures to prevent the escalation of anticipated operational occurrences and to maintain the effectiveness of the physical barriers placed between a radiation source or radioactive material and workers, members of the public, or the environment, in operational states and, for some barriers, under accident conditions. The objectives of defence in depth are (a) to compensate for potential human and component failures, (b) to maintain the effectiveness of the barriers by averting damage to the facility and to the barriers themselves, and (c) to protect workers, members of the public and the environment from harm under accident conditions in the event that the barriers are not fully effective.

1.3. The Safety Requirements publication on the Safety of Research Reactors [1] establishes the basic requirements in all the relevant areas associated with the safety of research reactors, including site evaluation, design, operation, decommissioning, regulatory oversight and the management system. This Safety Guide provides recommendations on fulfilling the basic requirements in all relevant areas associated with the safety of research reactors. In making use of this publication and the other Safety Guides that provide recommendations on means of fulfilling the basic requirements established in Ref. [1], the particular characteristics of research reactors should be taken into account. These characteristics include the wide variety of designs, the wide range of power levels, the different modes of operation and purposes of utilization, the particularities of siting and the differences among the operating organizations for research reactors. These characteristics require flexibility in setting objectives and applying the basic requirements when dealing with some specific topics.

1.4. The organizations involved in ensuring the safety of research reactors and the protection of site personnel, the public and the environment have a number of responsibilities that are interrelated. Some of the more important of these are the performance of the safety analysis and the development of the OLCs and safety related documents for review and assessment by the regulatory body. Documented operating procedures having safety significance are part of such documentation. The OLCs are a set of operating rules that normally includes: safety limits and safety system settings on relevant variables and parameters of the reactor; limiting conditions on equipment and operational characteristics of the reactor; surveillance requirements; and administrative requirements. This set should satisfy the basic requirements for the OLCs as established in Ref. [1], paras 7.29–7.41.

1.5. There is an international consensus on the need for the establishment of and compliance with OLCs for the safe operation of nuclear reactors. However, the OLCs are presented in various formats, depending on the structure of the mandatory safety documentation as stipulated in the legal framework established by the State for the operation of research reactors. Thus, some States include the OLCs, together with requirements of other kinds, as an essential part of the reactor licence for operation, while others make reference to a separate document containing the OLCs. In this regard, the structure of the present Safety Guide is oriented to the development of the OLCs as a separate document, which may or may not be attached to the reactor licence and appropriately referenced in the safety analysis report. In the case that the safety analysis report does not include the OLCs, a summary should be

presented in the safety analysis report and reference should be made to the separate document. This summary should be prepared in accordance with the recommendations provided in Ref. [2], paras A.1701–A.1708.

OBJECTIVE

1.6. The purpose of this Safety Guide is to provide recommendations on all important aspects of developing, formulating and presenting the OLCs and the operating procedures for research reactors. In particular, detailed recommendations are provided on the development, content and implementation of OLCs and operating procedures for the operation of research reactors as well as for experiments (see Ref. [3], para. 303). This Safety Guide is directed at both operating organizations and regulatory bodies.

SCOPE

1.7. This Safety Guide covers the concept of OLCs, their content as it applies to research reactors, and the responsibilities of the operating organization in respect of their establishment, modification and documentation, and with regard to compliance with them. The operating procedures necessary to facilitate full compliance with the OLCs and to ensure their observance are also within the scope of this Safety Guide.

1.8. The recommendations provided in this Safety Guide are intended to be applicable to most types of research reactor having power levels of up to a few tens of megawatts. Research reactors of higher power levels and specialized reactors (e.g. homogeneous reactors, fast reactors) may require the provision of additional guidance that is beyond the scope of this Safety Guide. Some flexibility in the application of the recommendations provided in this Safety Guide is warranted on the basis of the specific characteristics of research reactors (see para. 1.11) or the relevant regulations of States. Because of the differences in the approaches to the preparation of OLCs and operating procedures in different States, the level of detail required in the application of the regulatory body to develop a set of OLCs acceptable to the regulatory body.

1.9. Research reactors are used for various purposes, including research, training, radioisotope production, neutron therapy, neutron radiography and

material tests, and therefore they have differing design features and operational regimes. Design and operating characteristics may vary significantly, since experimental devices may affect the performance of reactors. In addition, the need for greater flexibility in the use of research reactors may require an approach to achieving or managing safety different from that for nuclear power plants.

1.10. Low risk research reactors having a power rating of up to tens of kilowatts and critical assemblies may need less comprehensive OLCs and operating procedures than those outlined here. While all recommendations in this Safety Guide should be considered, some may not apply to these low risk reactors. For these reasons, the recommendations and guidance provided in this Safety Guide should be graded for applicability to a particular research reactor (see Ref. [1], paras 1.11–1.14).

1.11. The factors to be considered in deciding whether certain recommendations provided here may be waived in applying a graded approach to the application of safety requirements include (but are not limited to):

- (a) Reactor power levels;
- (b) The radiological source term;
- (c) The amount and enrichment of fissile and fissionable material;
- (d) Spent fuel elements, high pressure systems, heating systems, fuel accumulators, and other systems and components that may affect the safety of the reactor;
- (e) The type of fuel element;
- (f) The type and mass of moderator, reflector and coolant;
- (g) The amount of reactivity that can be introduced and its rate of introduction, reactivity control, and inherent and engineered safety features;
- (h) The quality of the containment or the means of confinement;³
- (i) Utilization of the reactor (experimental devices, tests, reactor physics experiments);

³ The means of confinement is a barrier surrounding the main parts of a reactor containing radioactive material that is designed to prevent or mitigate the uncontrolled release of radioactive material to the environment in operating states or in design basis accidents. If the design is also able to fulfil this function under overpressure conditions after an accident, it is often called a containment.

- (j) The presence of personnel in the installation during the operation of the reactor;
- (k) The type of operation of the facility (e.g. whether permanent or only for short experiments);
- (1) Siting of the reactor;
- (m) Proximity of the reactor to population groups.

1.12. The OLCs represent an envelope of parameters. Compliance with the OLCs will protect the reactor, and will protect the staff, the general public and the environment from undue radiation exposure. In addition, establishing the OLCs forms an important part of meeting the requirements for the issuing by the regulatory body of the authorization to operate the facility. Therefore the requirement to develop the OLCs and to operate the facility in accordance with them should not be waived. However, the number of OLCs may vary depending on the factors listed in para. 1.11.

1.13. The development and use of operating procedures make a major contribution to the safe operation of the reactor. The operating procedures facilitate full compliance with the OLCs, and therefore the preparation and use of operating procedures should not be waived in any case. Since the operating procedures are required to be consistent with the OLCs, the number and the complexity of the procedures may vary depending on the factors listed in para. 1.11.

STRUCTURE

1.14. This Safety Guide consists of eight sections and two appendices. Section 2 describes the relationship between the OLCs and the fundamental safety objective of protecting people and the environment from harmful effects of ionizing radiation, and discusses: the concept of OLCs; the need for OLCs; their development; the roles and responsibilities of the operating organization and the regulatory body in their preparation and review; and the presentation of each operational limit or condition. Section 3 discusses the content of the OLCs document⁴ and provides a list of the safety parameters and systems that

⁴ OLCs are a set of rules setting forth parameter limits, and the functional capability and performance levels of equipment and personnel approved by the regulatory body for safe operation of an authorized facility. The document that contains the OLCs as a whole is referred to as the 'OLCs document'.

should be covered by the OLCs, including surveillance requirements for these parameters. This section also provides guidance on administrative requirements that should be covered by the OLCs. Section 4 covers the development of operating procedures. It describes the functions and responsibilities of the operating organization, operating personnel, the safety committee, the reactor manager and the regulatory body in developing and implementing procedures, and discusses general considerations regarding their categorization and other issues. Section 5 provides guidance on the format and content of procedures of various types to achieve consistency and completeness in developing the operating procedures. In addition, it describes specific topics to be addressed in procedures of each category. Section 6 provides guidance on the training of personnel in the use of procedures. Section 7 provides guidance on how to ensure compliance with OLCs and operating procedures, including the need to retain records of such compliance. Section 8 provides guidance on quality assurance pertaining to the OLCs and on the quality assurance practices applied to the development and implementation of the operating procedures. Appendix I provides a list of selected factors that should generally be considered in establishing the limiting conditions for safe operation and the surveillance requirements. Appendix II provides a list of typical procedures for the categories identified in Section 5.

2. DEVELOPMENT OF OPERATIONAL LIMITS AND CONDITIONS

GENERAL

2.1. To prevent situations from arising that might lead to accident conditions and to mitigate the consequences of accident conditions if they do arise, "A set of OLCs important to reactor safety, including safety limits, safety system settings, limiting conditions for safe operation, requirements for inspection, periodic testing and maintenance and administrative requirements, shall be established and submitted to the regulatory body for review and assessment" (Ref. [1], para. 7.29).⁵

⁵ See also para. 510 of the IAEA Safety Fundamentals publication on The Safety of Nuclear Installations, Safety Series No. 110, IAEA, Vienna (1993).

2.2. Thus the OLCs form an envelope or boundary for reactor parameter values and system conditions, within which the operation of the reactor has been demonstrated in the safety analysis report to be safe, and the site personnel, the public and the environment are adequately protected against radiation hazards. Thus the OLCs contribute to the prevention of accidents and to mitigation of the consequences of accidents if they do occur.

2.3. The OLCs are required to be developed to ensure that the reactor facilities are operated in accordance with the design assumptions and intent. In order to achieve this requirement, the safety analysis report should be developed so as to identify clearly the OLCs that must be met.

2.4. The OLCs should be submitted for the review and approval of the regulatory body as an important part of the licensing process, on the basis of which the operating organization is authorized to operate the facility.

2.5. The important function of the OLCs in the operation of the research reactor requires that they be adequately selected, clearly established and appropriately substantiated by means of a written statement of the reasons for their adoption. When reviewed and approved by the regulatory body, the OLCs represent an accepted basis for the operation of the reactor facility.

2.6. The technical aspects of the OLCs cover the limitations to be observed, as well as the operational requirement that structures, systems and components important to the safety of the research reactor be able to perform their intended functions as assumed in the safety analysis report. Safe operation depends on operating personnel as well as on equipment; OLCs should therefore also cover actions to be taken and limitations to be observed by operating personnel.

2.7. Regarding the functions of operating personnel, the OLCs include those principal requirements for surveillance and corrective or complementary actions that are necessary to supplement the functioning of equipment involved in maintaining the established OLCs. Some OLCs may involve combinations of automatic functions and actions by operating personnel.

2.8. The OLCs at research reactor facilities should include the following items:

- (a) Safety limits (see paras 3.5–3.13);
- (b) Safety system settings (see paras 3.14–3.18);

- (c) Limiting conditions for safe operation (see paras 3.19–3.26);
- (d) Surveillance requirements (see paras 3.27–3.32);
- (e) Administrative requirements (see paras 3.33–3.43).

Each OLC should be supported by a clear statement of its objective, applicability, specification and justification, as appropriate. These items should be included in the documentation on OLCs to increase the facility personnel's consciousness of their application and observance.

2.9. The development of the OLCs is based on the reactor design, on the safety analysis and on the information in the safety analysis report concerning the conduct of operations. The style and the format of the OLCs should be appropriate for their main purpose (see para. 2.1), and the following aims should also be taken into account:

- (a) To facilitate verification that operation is in compliance with the approved OLCs;
- (b) To facilitate understanding and awareness by the operating personnel of the application of and need for compliance with the OLCs.

2.10. The operating organization is required to be responsible for the preparation of the OLCs and for their submission to the regulatory body, as one of the requirements for the granting of a licence (see paras 3.11(f) and 7.29 of Ref. [1]). The operating organization should consult the designer in preparing the OLCs and should ensure that the operating personnel know the OLCs and adhere to them. In addition, the proposed OLCs should be reviewed by the safety committee⁶ before their submission to the regulatory body, the operating organization should ensure that the approved OLCs are revised appropriately.

2.11. The operating organization should prepare OLCs for each stage of reactor operation that may require a regulatory licence. For example, the commissioning stage of the reactor usually requires specific OLCs, which may be revised after completion of this stage. Similarly, operation of the reactor under special conditions, such as the conduct of a particular experiment or the undertaking of a reactor modification, may require special OLCs. Other reasons for a change in the OLCs may be an observed inadequacy of present

⁶ See Ref. [1], para. 4.15.

parameter values or requirements, experience gained during reactor operation, technological progress, and extended shutdown or decommissioning. The OLCs are therefore normally reviewed and changed as necessary over the reactor's lifetime.

2.12. In line with its overall responsibility, the operating organization is responsible for conducting a periodic review of the OLCs, in association with a review of the safety analysis report, so as to make revisions on the basis of operating experience. Accordingly, the operating organization is also responsible for the timely submission of any additions or changes to the existing OLCs to the regulatory body for review and approval.

2.13. In accordance with para. 2.10, the operating organization is responsible for ensuring compliance with the approved OLCs. To assist in discharging this responsibility, safety limits, safety system settings and surveillance requirements are required to be established in the OLCs (see Ref. [1], para. 7.29). In this context, the operating organization is responsible for keeping adequate records to facilitate audits and inspections to verify that the operating organization is responsible for setablishing procedures to be followed in the event of a violation of a safety limit or limiting condition.⁷

2.14. The operating organization should submit the OLCs to the regulatory body for review, assessment and approval⁸. The objective is to verify that each OLC is well founded, provides an adequate safety margin in relation to accidents analysed in the safety analysis report and complies with the applicable regulations.

2.15. The regulatory body should conduct regulatory inspections of the operating organization, reactor management and operations of the reactor facility to verify compliance with the approved OLCs. Further guidance on regulatory inspections is provided in Ref. [4].

⁷ See paras 7.39–7.41 of Ref. [1], and paras 7.3 and 7.4 of this Safety Guide.

⁸ Approval is not required in some Member States.

ATTRIBUTES OF OPERATIONAL LIMITS AND CONDITIONS

2.16. The presentation adopted for the OLCs may vary from State to State depending on national regulations and practices and on the particular reactor. It may range from a short list of limits and limiting conditions to a set of detailed specifications together with the objective, applicability statement and basis (see para. 2.8) for each of the specifications. The facility personnel should be made fully aware of the importance of OLCs and compliance with them. The latter format, consisting of a brief description of the objective, applicability, specification and basis for each of the safety limits, safety system settings and limiting conditions for safe operation, should be given consideration as a good practice. The presentation may also include a statement describing actions to be taken, with the allowed completion time, in the event of deviations from the established OLCs (see para. 3.20) or in the case of violation of an operational limit or condition (see para. 3.43).

2.17. The presentation adopted for the OLCs may also be used in an appropriate manner for surveillance requirements (see paras 3.27–3.32) and administrative requirements. The surveillance requirements may be included in the section of the document on the limiting conditions for safe operation or may be specified in a separate section of the OLCs document. The surveillance requirements should include the requirements for inspection, operability checks and calibrations, as applicable, and should clearly establish the frequency and scope of the tests required to verify that the performance levels for safe operation as set in the OLCs are met.

2.18. Clear presentation and avoidance of ambiguity are important factors in reliability in the use of OLCs, and advice on human factors should be sought at an early stage in the development of the documentation in which the OLCs will be presented to the operating personnel. The meaning of terms should be explained to help prevent misinterpretation.

2.19. Whenever modifications to the OLCs become necessary, the approach outlined in para. 2.18 should be followed. All modifications to the facility should be reviewed to determine whether they necessitate modifications to the OLCs. Any modification to the OLCs should be subject to review by the safety committee and to assessment and approval by the regulatory body, as required [3].

2.20. Whenever it is necessary to modify the OLCs on a temporary basis, for example, to perform tests or experiments, particular care should be taken to

ensure that the effects of the change are analysed. The modified state, although temporary, necessitates at least the same level of approval as a permanent modification. Any reasonable alternative approach, if available, should be preferred to the temporary modification of an OLC.

2.21. Periodic reviews should be undertaken to ensure that the OLCs remain applicable for their intended purpose, and, where necessary, the OLCs should be modified in the light of operating experience and technological developments. This periodic review, in association with a review of the safety analysis report, should be carried out even if the facility has not been modified.

2.22. Consideration may be given to the application of probabilistic safety assessment in the optimization of OLCs. Probabilistic assessment methods together with operating experience may be used in the justification and modification of OLCs.

Objective of the specification of OLCs

2.23. The OLCs should be meaningful to the responsible operating personnel. The objective of the specification of the OLCs should be clearly stated, and the OLCs should be specified in terms of measurable or directly identifiable values of parameters. This is important because the objective may not be evident from the specification itself. For example, if the objective of a particular operational limit or condition is stated as to ensure the integrity of the fuel cladding, a temperature may be specified. However, if there is no instrument for measuring the cladding temperature, it may be necessary to specify a reactor power level, coolant flow rate through the core, coolant inlet temperature and height of water above the fuel. It is not self-evident from these four specifications that the final objective is to ensure the integrity of the cladding. The relationship of a limiting parameter with other measurable parameters should be indicated by means of tables or diagrams, as appropriate. The limit or condition should be stated in such a way that it is clear whether a violation has or has not occurred in any situation.

Applicability of the specification of OLCs

2.24. The applicability of an operating limit or condition is specified in a statement that indicates the operational state (e.g. startup, normal operation, refuelling), the variables, the components, the systems and the administrative requirements to which the specification applies. Such a statement of applicability should be included to ensure a clear and proper understanding of

the scope of the specifications; for example, the specific cooling modes (natural, forced, etc.) in which a given limit is applicable may be specified.

Statement of the specification of OLCs

2.25. The specification of the OLCs provides a statement of the value of a particular parameter or the values of a group of parameters, either as a single value or as a range of possible values. The specification may concern a structure, a system, a component, an operation, or a surveillance requirement or administrative requirement. The specification should be stated in a clear and concise manner and should not conflict with other specifications. Specifications may be derived from the design, from the safety analysis report and from experience.

Bases for the specification of OLCs

2.26. Reasons should be given for the selection of the specification. These reasons should be based on the safety analysis report, on the reactor design or on aspects relating to the conduct of operations. These reasons or bases may be simple but conservative statements made on the basis of operational experience or experimental results. Reference to the relevant sections of the safety analysis report with a brief summary should be included in the bases.

2.27. The bases should show that the specification is conservatively selected for normal operation. Appropriate consideration should be given to factors such as calibration errors, measurement accuracy and system response times or operator response times.

3. CONTENT OF THE OPERATIONAL LIMITS AND CONDITIONS DOCUMENT

3.1. Just as the format of the OLCs may differ between States, so may the content and the order of presentation. However, all items relevant to safe operation should be included in the OLCs. The content and presentation of the OLCs document may be arranged in the following manner: table of contents; definitions; introduction; safety limits; safety system settings; limiting

conditions for safe operation; surveillance requirements; administrative requirements.

TABLE OF CONTENTS

3.2. The table of contents should provide sufficient detail to allow for easy reference to a specific operational limit or condition.

DEFINITIONS

3.3. Definitions of specific terms used in the OLCs document should be given. In addition, the definitions of terms particular to the research reactor should be provided. (See the definitions in the IAEA Safety Glossary [5].)

INTRODUCTION

3.4. The introduction should contain general information about the research reactor operating organization and the authorship of the OLCs document, including its historical development, if necessary. It should also include a statement of any restrictions placed on the OLCs. An example is: "These OLCs are applicable during the commissioning stage of the reactor only". Finally, there should be a statement that all operations of the reactor should be in accordance with the OLCs.

SAFETY LIMITS

3.5. Safety limits are limits on process variables within which the operation of the research reactor facility has been shown to be safe. Safety limits are necessary to protect the integrity of the principal physical barrier that guards against uncontrolled radioactive releases in all operational states and design basis accidents.⁹ For many research reactors, this principal physical barrier is the cladding of the fuel elements, whose temperature is maintained below a

⁹ Those safety limits that might be exceeded in the event of a design basis accident should be justified on the basis of careful analyses.

certain limit by cooling so that the integrity of the cladding is ensured. For some reactors, the principal physical barrier is the primary coolant boundary.

3.6. The safety limits should be established by means of a conservative approach that ensures that all the uncertainties of the safety analyses are taken into account. This implies that the exceeding of a single safety limit does not always lead to unacceptable consequences. Nevertheless, if any safety limit is exceeded, the reactor should be shut down and normal operation should be restored only after an appropriate evaluation has been performed and approval for restarting has been given in accordance with established procedures.

3.7. The fuel temperature may be used to establish a safety limit if there is a provision for measuring it. If the temperature is measured in only one location in the core, the measured temperature should be correlated to the maximum fuel temperature in the core.

3.8. In some research reactors there may be no provision for measuring the fuel temperature. In this case, the safety limit is often expressed in terms of other related parameters that are measured, such as the thermal power level, coolant flow through the core, coolant inlet temperature or outlet temperature, coolant pressure and height of coolant above the core. If the core can be cooled by either forced or natural convection, safety limits should be developed for each mode.

3.9. Selection of the safety limits is of paramount importance and should be given careful consideration. For example, the onset of nucleate boiling, which is often used to establish safety limits, represents an undesirable but not unsafe condition for the reactor. Departure from nucleate boiling and flow instability, however, are conditions that, if approached too closely, would have significance for safety, and these conditions may therefore be used to establish safety limits. For this reason, reactor operation is limited to a power level such that the maximum heat flux in a fuel element is only a fraction of the burnout heat flux. In some instances (e.g. for low power research reactors), the safety limits may be set very conservatively.

3.10. Similarly, the maximum allowable surface temperature for the cladding should be set as a safety limit. This safety limit should be applied to the hottest reactor channel, and it should not be exceeded even during pump failure accompanied by reactor shutdown. It should be noted that some research reactors that are cooled by forced convection utilize downward coolant flow. In the event of pump failure and reactor shutdown, the reactor may be cooled by

natural convection, which involves a reversal of the direction of flow. The safety limit should not be exceeded even in such cases.

3.11. It should be shown in the safety analysis report or other associated document that the safety limits are not violated in any operational state.

3.12. The specification of the safety limit should be clear and precise, and the parameters to which it applies and the objective of the limit should be stated. The basis for the safety limit should provide sufficient information to allow a clear understanding of its safety significance on the part of the operating personnel and the regulatory body.

3.13. Although the integrity of the containment or the means of confinement (if any) is important in limiting the radiological consequences of an accident, a loss of containment or loss of the integrity of the means of confinement does not in itself lead to damage to the fuel cladding. The integrity of the containment or the means of confinement is therefore not included in the safety limits, but it should be included under the limiting conditions for safe operation.

SAFETY SYSTEM SETTINGS

3.14. For each parameter for which a safety limit is required, and for other important safety related parameters¹⁰, the safe design of the reactor requires the installation of a system that monitors the parameter and provides a signal that can be utilized in an automatic mode to prevent the value of that parameter from exceeding the safety limit. The set point for this protective action is defined as the safety system setting.

3.15. Some safety system settings are provided to initiate the operation of engineered safety systems to limit the course of anticipated operational occurrences in such a way that either safety limits are not exceeded or the consequences of postulated accidents are mitigated.

3.16. Established safety system settings should ensure the automatic actuation of safety systems within the parameter values assumed in the safety analysis

¹⁰ For example, a limiting condition for safe operation may be the concentration of radioactive material in the exhaust ventilation, which requires monitoring and action to prevent releases exceeding a set value.

report, despite possible errors that could occur in adjusting the nominal set point. Appropriate alarms should be provided to enable the operating personnel to initiate corrective actions before safety system settings are reached.

3.17. Safety system settings should be established for all operational states of the reactor. In determining a safety system setting, the process uncertainties and measurement uncertainties, the response of instrumentation and uncertainties in calculations should be taken into account (see Ref. [1], para. 7.34).

3.18. The specification of any safety system setting should be clear and precise, and should state the parameters to which it applies and the objective of the specification. The basis for the safety system setting should contain sufficient information to allow a clear understanding of its safety significance on the part of the operating personnel and the regulatory body.

LIMITING CONDITIONS FOR SAFE OPERATION

3.19. Limiting conditions for safe operation are administratively established constraints on equipment and operational parameters that must be adhered to during the startup, operation, shutting down and shutdown of a research reactor to provide acceptable assurance of safe operation within the reactor licence conditions and within applicable regulations. The limiting conditions for safe operation should be consistent with and, to the extent possible, should be derived from the safety analysis report. Where established, limiting conditions for safe operation should be complied with in all operational states of the reactor (see Ref. [1], para. 7.35). Compliance with limiting conditions for safe operation prevents safety system settings from being reached, addresses other factors that, if not controlled, might give rise to risks to the health and safety of the public, and ensures full capability for performing other necessary safety functions in all operational states and under accident conditions.

3.20. The limiting conditions for safe operation should include limits on operating parameters, requirements for minimum operable equipment with systems unavailability rules and the time allowed to attain the safety fallback state in the event that these unavailability rules are not met, and prescribed actions to be taken by the operating personnel in the event of deviations from the established OLCs as well as the time allowed to complete these actions.

3.21. Operability requirements should state for the various modes of normal operation the number of systems or components important to safety that should be either in operating condition or in standby condition. These operability requirements define the minimum safe facility configuration for each mode of normal operation. The actions to be taken in the case that operability requirements cannot be met to the extent intended should be specified, and the time allowed to complete these actions should also be stated.

3.22. Given the higher associated risk during startup of the reactor, the operability requirements for this mode should be more stringent than those set for purposes of operational flexibility in power operation. Systems, structures and components that are required to be operable for startup should be specified.

3.23. For the operability requirements of safety related equipment, the provisions in the design for redundancy, the reliability of the equipment and the period over which equipment may be allowed to be inoperable without an unacceptable increase in risk should be taken into consideration. The basis for determining an unacceptable increase in risk should be clearly documented in the OLCs.

3.24. The objective and applicability of each limiting condition should be stated, its specification should be clear and precise, and its basis should contain sufficient information for it to be fully understood by the operating personnel and easily assessed by the regulatory body.

3.25. The number of limiting conditions for safe operation may be large, even for a low power research reactor. For this reason the limiting conditions should be grouped by topic. An example of one such grouping is as follows:

- (a) Fuel, fuel elements and assemblies;
- (b) Fuel handling and storage of fresh and spent fuel;
- (c) Reactor core configuration;
- (d) Reactivity and reactivity control systems;
- (e) Protection systems and reactor shutdown systems;
- (f) Fuel loading, reactor startup and operation;
- (g) Coolant systems and connected systems;
- (h) Containment systems or means of confinement, including ventilation;
- (i) Operational radiation protection;
- (j) Instrumentation and control systems;
- (k) Experimental devices;

- (1) Electric power supply systems;
- (m) Auxiliary systems and equipment;
- (n) Other limitations.

3.26. A list of selected factors to be considered when establishing limiting conditions for safe operation is provided in Appendix I.

SURVEILLANCE REQUIREMENTS

3.27. To ensure that safety system settings and limits and conditions for safe operation are observed at all times, the relevant systems and components should be monitored, inspected, checked, calibrated and tested in accordance with an approved surveillance programme.

3.28. It should be an important design requirement for research reactors to allow for appropriate functional testing and inspection of all items important to safety (see para. 6.43 of Ref. [1]). This means that all items subject to safety system settings and limiting conditions for safe operation should undergo some form of surveillance or testing. Surveillance requirements should specify the frequency and scope of tests and the acceptance criteria (see paras 3.18–3.23 of Ref. [6]) to show that the performance requirements associated with the items subject to safety system settings and limiting conditions for safe operation of the surveillance frequency should give average intervals with a minimum frequency that is to be achieved or a maximum interval that is not to be exceeded in order to provide operational flexibility in scheduling the inspection, operability check or calibration.

3.29. The surveillance requirements should be specified in procedures with clear acceptance criteria so that there can be no doubts concerning system operability or component operability. The relationship between these acceptance criteria and the limit or condition being confirmed should be available in written form.

3.30. The surveillance requirements may be presented either by grouping them according to the systems to which they refer or in a mixed way; for example, in

¹¹ Surveillance requirements are met by using procedures that are normally part of the maintenance, periodic testing and inspection programme (see Section 5).

groups of requirements for major systems (such as the reactivity control system, the reactor pool or tank and the containment and/or confinement systems) or in groups of requirements covering related activities (e.g. tests before startup, monthly tests, quarterly tests, routeing, monitoring and personal dosimetry).

3.31. Limiting conditions for safe operation concerning the surveillance requirements should also cover activities to detect ageing and other types of deterioration due to corrosion, fatigue and other mechanisms. Such activities will include non-destructive examination of passive systems and of systems explicitly covered by limits and conditions for safe operation. If degraded conditions are found, the effect on the operability of systems should be assessed and acted upon appropriately.

3.32. Some of the OLCs, including surveillance requirements, may not apply during extended periods of shutdown of the reactor. For example, calibration of a power measuring channel may be deferred but should be performed before the next reactor startup. Some additional special surveillance requirements may be necessary during an extended shutdown period, such as during major maintenance or modifications. For these reasons, the surveillance requirements for extended shutdown periods may be specified separately.

ADMINISTRATIVE REQUIREMENTS

General

3.33. This section of the OLCs document specifies the administrative requirements for the reactor facility.¹² These administrative requirements consist of administrative controls concerning organizational structure and responsibilities, minimum staffing requirements, and required actions following violation of an operational limit or condition.

3.34. A list of safety related procedures could also be included in the OLCs document. These safety related procedures should be reviewed by the safety committee and may be subject to approval by the regulatory body.

¹² In some Member States, administrative requirements may be part of the quality assurance programme.

Organization

3.35. The organizational structure of the facility should be presented in an organizational chart with a brief description of functions. The chart should show the key personnel of the operating organization who have responsibility for the facility under the terms of the licence, including the reactor manager, shift supervisor and reactor operators.

Staffing

3.36. It should be an important requirement relating to operational safety to determine and specify the minimum staffing requirements of the various disciplines for all operational states of the reactor. These requirements will vary with the complexity and power level of the reactor. The minimum qualifications for key operating personnel should also be specified.

Training and retraining

3.37. The administrative requirements should include a statement that shift supervisors, reactor operators, radiation protection staff, experimenters, maintenance personnel and others who frequent the reactor facility should be properly trained. The staff requiring certification or licensing should be specified. If appropriate, the period of validity of the certification or licence should also be stated.

Review

3.38. The requirement for specified reviews by a safety committee¹³ should be stated. The items that should be reviewed by the safety committee may include the following:

- (a) Proposed changes to the OLCs or the facility licence;
- (b) Proposed changes to existing tests, experiments, equipment, systems or procedures, and new tests, experiments, equipment, systems or procedures having safety significance;
- (c) Safety related modifications to the reactor facility;
- (d) Violations of the OLCs, the licence conditions and procedures having safety significance;

¹³ See paras 7.25–7.26 of Ref. [1].

- (e) Events that are required to be reported, or have been reported, to the regulatory body;
- (f) Routine radioactive releases and exposures of personnel and the public;
- (g) Periodic reviews of the operation and safety performance of the facility.

Utilization and modifications

3.39. Administrative requirements for the safe utilization and modification of the facility should be included in the OLCs. Guidance for deciding which experiments or modifications have to be referred to the regulatory body should be included in the administrative requirements. Further guidance on utilization and modification of the facility is provided in Ref. [3].

Records and reports

3.40. An administrative requirement for the preparation and availability of various records and reports should be included in the OLCs.

3.41. The administrative requirements should state that the operating organization should make periodic summary reports to the safety committee and, if required, to the regulatory body on matters relating to the safety of the facility. The content and frequency of reports should be specified.

3.42. Records that are important for the safe operation of the reactor and for demonstrating compliance with the OLCs should be prepared and retained (see Ref. [1], para. 7.83). The records to be maintained by the facility and the time period for which the records are required to be retained should be specified. Typical records to be retained include the following:

- (a) The safety analysis report for the facility and changes to the safety analysis report;
- (b) The facility licence, the licence conditions and the OLCs;
- (c) Required plans such as the emergency plan, the security plan, the quality assurance plan, and the reactor operator and staff training plan;
- (d) Records of releases of effluents to the environment;
- (e) Records of radioactive waste;
- (f) Records important to decommissioning, such as records of spills, as-built drawings and records of modifications to structures;
- (g) Records of radiation exposures;
- (h) Records of significant contamination events;
- (i) Facility drawings;

- (j) Fuel receipt, shipment and inventory records;
- (k) Routine operating data such as log books and recording charts;
- (1) Procedures and changes to procedures;
- (m) Records of events that are required to be reported to the regulatory body;
- (n) Records relating to reactor experiments, such as application forms and collected data;
- (o) Facility radiation records and contamination survey records;
- (p) Records pertaining to the safety committee, such as meeting minutes and review reports;
- (q) Surveillance and maintenance records.

Action required in the event of a violation of OLCs

3.43. Actions that are required to be taken by the operating organization in the event that a safety limit or limiting condition for safe operation is violated or cannot be met should be included in the OLCs.¹⁴

4. DEVELOPMENT OF OPERATING PROCEDURES

GENERAL

4.1. All safety related activities should be performed in conformance with documents issued in accordance with approved operating procedures. The availability and correct use of written operating procedures is an important contribution to the safe operation of a research reactor. The IAEA Safety Requirements publication on the Safety of Research Reactors (Ref. [1], para. 7.51) states that "Operating procedures shall be developed for all safety related operations that may be conducted over the entire lifetime of the facility", and this includes normal operation, anticipated operational occurrences and emergencies.

4.2. The operating procedures for normal operation should be developed to ensure that the research reactor is operated in compliance with the OLCs and should provide instructions for the safe conduct of operation in all modes, such

¹⁴ See paras 7.39–7.41 of Ref. [1].

as startup, shutdown, power level changes, fuel loading and handling, normal operation and anticipated operational occurrences.

4.3. In this Safety Guide the term 'operating procedures' also covers other activities that are conducted by means of written instructions and step by step descriptions of operations, such as commissioning, maintenance, experiments and modifications, activities in emergencies and radiation protection services. Operating procedures are sometimes bound in several volumes called manuals (e.g. operation manual, testing manual, maintenance manual). For low power research reactors, the operating procedures may be collected into a single volume under the general title of operating instructions.

4.4. The organizations involved in ensuring the safety of research reactors and the protection of site personnel, the public and the environment have a number of responsibilities that are interrelated. Among these responsibilities are the performance of the safety analysis as well as the preparation of other safety related documents for review and assessment by the safety committee and approval by the regulatory body, as required. Operating procedures having safety significance are part of such documents.

4.5. Operating procedures should be verified and validated by authorized persons to ensure that they are administratively and technically correct, are easy to understand and to use, and will function as intended. Special attention should be paid to ensuring that operating procedures are compatible with operation in the environment in which they are intended to be used. The operating procedures should be validated in the form in which they will be used.

ROLES AND RESPONSIBILITIES

Operating organization

4.6. To achieve the objective of safe operation of a reactor, the operating organization should establish administrative controls. These controls will usually be in the form of procedures to be followed by all personnel involved in their implementation.

4.7. The operating organization is responsible for establishing a set of OLCs, including administrative and organizational requirements. These general

operating rules¹⁵ should be supplemented by specific written operating procedures.

4.8. When possible, operating procedures, especially for a new reactor, should be developed in cooperation with the reactor designer, equipment manufacturers and vendors. The assistance of external consultants may also be sought.

Operating personnel

4.9. The operating personnel may initiate and should participate in the process of developing the operating procedures.

4.10. The operating personnel should be knowledgeable about the application of the operating procedures relevant to their tasks in the facility.

4.11. The operating personnel should operate the reactor in accordance with valid operating procedures and should provide feedback to the reactor manager on the application of the procedures.

4.12. Those procedures that are directly related to radiation protection should be reviewed by a radiation protection officer. The radiation protection manager is responsible for the preparation of all the radiation protection procedures for the tasks carried out by radiation protection personnel.

Safety committee

4.13. The safety committee should review and assess the OLCs and the submitted operating procedures, and make recommendations before submission to the regulatory body for approval, as required. The safety committee should establish a process for review of urgently needed new procedures or changes to existing procedures that cannot await a review during a regularly scheduled safety committee meeting. For example, minor modifications to the operating procedures may possibly be made with the approval of the reactor manager followed by a review by the safety committee at its next meeting, provided that the general operating rules are observed.

¹⁵ These general operating rules are frequently compiled in a document called an operation manual.

Reactor manager

4.14. In addition to those procedures required by the regulatory body (see para. 4.19), the reactor manager should determine the need for operating procedures that may be initiated by the operating personnel or others.

4.15. The reactor manager should ensure the development and implementation of the operating procedures. Alternatively, the reactor manager may appoint a staff member, normally a relevant group leader, to develop procedures.

4.16. The reactor manager is responsible for the approval of all operating procedures, including those reviewed by the safety committee.

4.17. The reactor manager is responsible for the training and retraining of staff in the procedures. The reactor manager should ensure that the latest revision of the procedures is used in the training or retraining.

4.18. The reactor manager should ensure that operational procedures are readily available for reference close to the point where the work is done or operations are carried out. Usually, one full set of operational procedures is kept in the control room and another full set in the reactor supervisor's office. There may be an additional selection of appropriate procedures kept in a location near relevant operations areas that are remote from the control room.

Regulatory body

4.19. The regulatory body should review, assess and approve the OLCs and the submitted operating procedures, as required. The regulatory body may require that some specific procedures be established.

4.20. The operating procedures should be available to the regulatory body for its consideration.

CATEGORIZATION AND LIST OF OPERATING PROCEDURES

4.21. Procedures should be developed for all safety related operations that may be conducted over the entire lifetime of the research reactor. For this reason it is convenient to categorize procedures according to their purpose and content. One such categorization is as follows:

- (a) Commissioning procedures;
- (b) Operational procedures;
- (c) Maintenance procedures;
- (d) Inspection, calibration and periodic testing procedures;
- (e) Radiation protection procedures;
- (f) Procedures for the authorization of operation, maintenance, irradiation or experiments;
- (g) Procedures for operator response to anticipated operational occurrences;
- (h) Emergency procedures;
- (i) Physical protection procedures;
- (j) Procedures for the handling of radioactive waste and control of radioactive releases;
- (k) Decommissioning procedures;
- (l) Procedures for utilization and for modification of the reactor;
- (m) Administrative procedures.

This list is not intended to be comprehensive. Some items may not be appropriate for low power facilities, and the list may be incomplete for high power facilities. Appendix II presents selected operating procedures listed on the basis of this categorization.

4.22. The categorization has been selected for consistency with the topics developed in the IAEA Safety Requirements publication on the Safety of Research Reactors [1]. Categorizations on the basis of other criteria may also be valid. A categorization system may be based on the review and approval route for the procedures; for example, safety related procedures (required and approved by the regulatory body), administrative procedures (reviewed by the operating organization) and other safety related procedures (reviewed by the safety committee and approved by the reactor manager). In other categorizations, only the area of application may be considered.

GENERAL CONSIDERATIONS FOR THE DEVELOPMENT OF OPERATING PROCEDURES

4.23. Paragraphs 4.24–4.29 present general considerations relating to the different types of procedures presented in para. 4.21. In addition, where possible, a reference is provided to IAEA publications that provide additional information.

4.24. The need for special quality assurance measures should be taken into account when developing procedures for operational activities that have an influence on or that relate to the following:

- (a) Reactivity and criticality;
- (b) Thermal safety;
- (c) Safety of experiments;
- (d) Repair actions;
- (e) Modification of existing systems or components;
- (f) New installations;
- (g) Manipulation of special components and radioactive material;
- (h) Surveillance of equipment;
- (i) Inspection programmes;
- (j) Steps for the approval of different safety related actions (e.g. replacements, repairs, modifications, new installations);
- (k) Accreditation of operating personnel and experimenters.

4.25. To develop a set of procedures for use in operation, a planned and systematic process should be applied. This may be facilitated by using a prescribed format and standard outline of the content to be incorporated.

4.26. Each procedure should be sufficiently detailed for a qualified individual to be able to perform the required activities without direct supervision, but the procedure should not provide a complete description of the facility processes involved.

4.27. The format of the procedures may vary, but the procedures should be developed in accordance with established requirements and recommendations relating to quality assurance.

4.28. Persons with appropriate competence and experience should be designated to prepare and verify procedures.

4.29. Human factors should be taken into account to develop safe, reliable and effective operating procedures. Consideration should be given to the layout, the general design of the facility, staffing requirements and operating experience at the research reactor concerned.

STEPS IN THE PREPARATION OF A PROCEDURE

4.30. Once the objectives of the procedure have been defined, the first step in the development of an operating procedure is to evaluate the possible methods and staff requirements for fulfilling the objectives of the procedure and to select the methods best suited for achieving the desired goal.

4.31. If it is possible and considered necessary, the methods selected should be simulated by the staff and the relevant technical and administrative personnel who will be performing the task covered by the procedure. A draft procedure should be used for the simulation. The simulation should cover all conceivable technical and human errors that could occur during the performance of the task. The procedures should be finalized on the basis of the results of the simulation.

4.32. If the procedure has significance for radiation protection, it should be reviewed by a radiation protection officer and should be modified as recommended.

4.33. The draft of the procedure may be improved by means of a review by staff members with experience in the subject of the procedure and by those staff members who will use the procedure.

4.34. The reactor manager should be responsible for a review of a draft of the procedure to determine whether the procedure is sufficiently detailed for its intended objectives and is consistent with other relevant procedures. Also, the reactor manager should ensure that implementation of the procedure would not violate any OLCs.

4.35. The reactor manager should forward the final drafts of safety related procedures to the safety committee for review and comment prior to their approval. Procedures having significance for the safety of the reactor should be specified as such and should be subject to approval by the regulatory body, as required.

4.36. The procedure may be released by the reactor manager for a trial period or a period of restricted use and subsequently revised if necessary. Following this validation period, use of the procedure should be subject to final approval by the reactor manager. 4.37. Experience gained in the use of a procedure may indicate a need for its review and updating. Review and updating should, in principle, follow the same steps as those in the preparation of a procedure as outlined here. The reactor manager should specify the process for this review and updating.

4.38. Safety significant revisions of procedures should be treated in accordance with para. 4.35. Other revisions of procedures may be approved directly by the reactor manager.

5. FORMAT AND CONTENT OF OPERATING PROCEDURES

GENERAL

5.1. All operating procedures should have a standard format as specified by the operating organization. They should follow a suitable sequential presentation, should have a clear, concise and logical text, and should cover all relevant issues. The following guidance will facilitate meeting these objectives.

5.2. To ensure consistency in format and content, operating procedures should be prepared in accordance with a general quality assurance procedure or administrative procedure that governs the development, review and control of such documents. Provision should be made in this general procedure or elsewhere for periodic review. There should be a mechanism to verify conveniently that a procedure has been approved (e.g. by means of a signature) and that it is current (e.g. through a list of the latest revision dates).

5.3. Operating procedures should generally direct the step by step performance of all activities except for routine activities that are capable of being performed by qualified personnel without special instructions. More than one procedure may be necessary to accomplish certain tasks. In this case, the order of performance of the procedures should be specified in the procedures themselves.

5.4. There should be a clear differentiation in a procedure between the introduction, guidance and essential steps.

5.5. The following is a typical example of the content to be considered for inclusion in an operating procedure:

- (a) Identification number: A unique number that identifies the procedure as one of a series of operating procedures.
- (b) Revision number and date: The current revision number and date should be included at the head of each page of an operating procedure to ensure that personnel are aware of the revision that is in use (e.g. Revision 3: 16 July 2003).
- (c) Date of expiry: This may be the date of the next revision or it may be indefinite until the withdrawal of the procedure.
- (d) Title: A concise description of the content of the procedure (e.g. reactor startup, control rod calibration).
- (e) Scope and purpose: A statement of the scope and purpose of the procedure.
- (f) Definitions: Definitions of terms used in the procedure that are necessary for understanding and performing the procedure.¹⁶
- (g) References: References on the basis of which the procedure was developed and that are referred to in the text.
- (h) Responsibilities: Specification of the requirements on and responsibilities of personnel performing the procedure.
- (i) Prerequisites: Specific conditions for the reactor, systems, equipment and personnel required to perform the procedure.
- (j) Additional requirements: Specification and/or description of special tools, support services, radiation protection measures, special safety precautions, possible effects on the reactor during the implementation of the procedure, measures to prevent damage to fuel and equipment, training (education) of personnel, conditions of applicability and preparations relating to the performance of the procedure.
- (k) Instructions: Specific step by step instructions for performing a procedure. The level of detail should be such that qualified personnel can follow the instructions without further directions. Where appropriate, hold points for inspection and verification should be specified. Instructional steps should begin with verbs (i.e. actions such as turn, record, energize, set, note, check, lift, press, test, insert). Caution should be exercised to ensure that the steps in the instructions are complete, and that all required actions have been considered and there are no instruction steps contained in the procedure as prerequisites and special

¹⁶ In general, use of terminology requiring definitions should be avoided. If this is not possible, a definition needs to be incorporated into the procedure.

requirements that could be overlooked while performing the procedure. Consideration should be given to setting up instructions in the form of checklists on which each step is initialled by operating personnel as it is completed. Where necessary, instructions should be provided for closing activities such as restoring the system to its normal configuration.

- (1) Basis for measurements and calculations: For some procedures that require measurements and calculations, documentation of the basis for the measurements and calculations may be provided to assist operating personnel in understanding the steps performed in the procedure.
- (m) Documentation: While completion of some procedures may be indicated by a simple entry in the reactor logbook, more complicated procedures may be documented by means of appropriate checklists, data tables and reports. Such data sheets should include the name of the person recording the data, the time and the date. The level of detail for documenting a procedure should be commensurate with the relative complexity of the operation and the inherent risks associated with failure to conduct the operation properly.

5.6. Procedures should include arrangements for collecting, tabulating and reporting data and test results. Methods of analysis should be stated and presented in a manner that allows for further verification. Test data should be evaluated against predefined performance parameters and acceptance criteria in which account is taken of the uncertainties assumed in the safety analysis.

5.7. On the basis of the quality assurance requirements, procedures should be prepared for the procurement and acceptance of new components or equipment (e.g. fuel elements, ion exchange resins). The procedures should require a valid calibration certification for test equipment that is used in the procedure.

5.8. Procedures should specify the authority for permitting deviations from the procedure and the circumstances under which deviations are permitted. Such deviations, if permitted, should be made within the bounds of the relevant OLCs.

5.9. In addition to the general recommendations provided here for consideration in developing all procedures, the following sections are intended to provide additional recommendations for the various types of operating procedure.

COMMISSIONING PROCEDURES

5.10. The content of commissioning procedures for tests of equipment and systems should be defined in accordance with para. 5.5, with special attention paid to the following:

- Summary of purpose, equipment to be tested and relationship to the commissioning programme;
- Prerequisites and initial conditions;
- Precautions, including stopping of the test;
- List of required equipment and instruments;
- List of data to be recorded and checklists;
- Analysis of data and results;
- Acceptance criteria;
- Provisions for corrective actions concerning possible non-conformances;
- Certification of completion of the test.

5.11. Commissioning procedures may be subdivided into those that are derived directly from operating procedures and those that will be required during commissioning only. The latter are sometimes referred to as test procedures.

5.12. A commissioning procedure should be prepared for each commissioning test or activity. The procedure may also be used as a guide for assessing and documenting the results of the test.

5.13. The test procedures should follow the normal facility operating procedures to the extent practicable, to verify these procedures and to provide an opportunity for operating personnel to become familiar with the normal facility operating procedures.

5.14. Certain commissioning activities may simply require generic procedures or lists of instructions.

5.15. Commissioning procedures should include prerequisites, where applicable, for system tests that must be completed prior to performing the step by step instructions established by these procedures, including commissioning of support systems (e.g. operable flow measurement channel prior to testing of the primary coolant system, operable electric power supply prior to commissioning tests of pumps).

5.16. In addition to operational procedures, supporting documentation, including manufacturers' manuals and construction drawings, may be required for commissioning test procedures for some components and systems.

5.17. If necessary, the procedure should include hold points for the notification and involvement of outside agencies, manufacturers and the regulatory body.

5.18. Procedures should state all the necessary changes to the normal operating configuration that are required for testing. However, in this case, configuration checks should be undertaken to ensure that these deviations are made correctly before the start of the tests and that all the components or systems are restored to their normal status after the testing.

5.19. Consideration should be given during the preparation of the procedure to interactions between the reactor and experimental devices.

5.20. Commissioning procedures should be prepared for experimental devices in any case, whether they will undergo commissioning at the same time as the reactor or later.

5.21. The commissioning procedure should contain provisions for dealing with unexpected results of commissioning, deliberate changes to the design, programmes or tests that may become necessary, and incidents that may occur during the commissioning process.

5.22. In practice, many of the commissioning procedures become operational procedures (e.g. procedures for fuel loading, startup, calibration of reactivity control mechanisms and determination of the thermal power level).

OPERATIONAL PROCEDURES

5.23. The operating organization should ensure that all reactor operation is in accordance with approved operational procedures. These procedures should be prepared for all activities performed by the operating personnel for all operational states of the reactor and, where appropriate, for experiments and for the handling of fuel assemblies or other core and reflector components, including experimental devices.

5.24. Normally, procedures are performed one at a time. If this is not the case, the safety significance of the simultaneous performance of several procedures

should be taken into account in the procedure by specifying special precautions.

5.25. Operating procedures are reviewed periodically and also whenever a change is made in the configuration of the reactor system or components concerned. For operations that are performed infrequently, the existing procedures should be reviewed in any case before use, and revisions should be made as appropriate.

5.26. Operational procedures should specify the actions required in the case of an unexpected event or unexpected results during the performance of the procedure, as necessary.

5.27. Operational procedures should include requirements for work permits, if necessary.

MAINTENANCE PROCEDURES

5.28. A plan for preventive maintenance should be prepared to assist in the timely completion of those maintenance activities that are required to be completed on a regular (e.g. weekly, monthly, quarterly, semi-annual, annual) basis.¹⁷ Procedures for corrective maintenance activities should be prepared as required.

5.29. In the preparation of maintenance procedures, particular attention should be paid to the effects of the procedure on safety systems and on reactor operation. Some procedures with no impact on reactor safety may be performed during reactor operation; other procedures may require shutdown of the reactor. The maintenance process should not reduce the safety of the reactor, and the OLCs should not be violated.

5.30. A system of work permits should be used for maintenance, including appropriate check-offs, during and after the conduct of the work in accordance with the quality assurance programme. This is to ensure that all work is conducted with the knowledge and permission of the person in operational

¹⁷ Ageing management activities are generally part of a preventive maintenance programme. Procedures include inspections to evaluate the effects of ageing of equipment and the need for the replacement of equipment.

control of the reactor and that both the safety of the reactor and the safety of the personnel doing the work have been considered. Therefore, maintenance procedures may incorporate the requirement of a work permit as a prerequisite to performing the maintenance.

5.31. Maintenance procedures should specify any changes to the normal reactor operating configuration (e.g. valve line-ups) and should have provisions for the restoration of the normal configuration after maintenance (e.g. procedures for taking mechanical and electrical equipment out of service and for restoring it to service). A generic procedure should therefore be developed or special provisions should be made in individual procedures to achieve configuration control.

5.32. Reference to drawings, manufacturer's manuals to be used and manufacturer's recommendations to be followed should be included in the procedures. It should be ensured that the updated versions of drawings and manuals are used.

5.33. Maintenance procedures should specify that the results of maintenance, testing and inspection should be assessed by properly qualified personnel. Comparison should be made, where appropriate, with the results of previous inspections and tests to determine potential failures and to permit timely corrective action.

5.34. Maintenance procedures should specify that the resumption of normal operation should be permitted only after an authorized person has approved the results of the maintenance.

5.35. Special procedures should be prepared to control contractor maintenance work. These procedures should include prerequisites to the work, requirements for supervision of the contractor, contractor qualification and work coordination.

INSPECTION, CALIBRATION AND PERIODIC TESTING PROCEDURES

5.36. Periodic testing is performed to fulfil surveillance requirements specified in the OLCs and is intended to ensure compliance with the OLCs. The surveillance programme should be adequately specified to ensure the inclusion of all aspects of the limits or conditions. 5.37. Periodic testing procedures, which include calibration, inspection and operability checks, should be prepared for structures, systems and components that are important for safe operation of the reactor. A plan for scheduling calibrations, inspections and operability checks should be prepared to assist in their completion at the required frequencies.

5.38. The frequency of the surveillance activities should be stated and should be based on:

- Analyses, including insights from probabilistic safety assessment, where available;
- The recommendations of the supplier;
- Experience gained from previous surveillance results;
- Engineering judgement.

5.39. Periodic testing procedures should be consistent with the reactor operating procedures. In addition, when developing periodic testing procedures, the considerations for developing operational and maintenance procedures (see paras 5.23–5.35) should also be taken into account.

5.40. Each periodic testing procedure should provide for a final acceptance (including signature) by a person qualified and authorized to assess the results of the procedure and to verify compliance with the OLCs.

5.41. Periodic testing procedures should have provisions for resolving non-conformances with the OLCs.

5.42. Acceptance criteria (for which consideration has been given to the uncertainties of measurements) should be provided within periodic testing procedures. Providing a range of acceptable values for parameters is therefore generally better than giving single values.

5.43. The procedures may require that instruments used for calibrations be certified in accordance with the requirements of the quality assurance programme.

RADIATION PROTECTION PROCEDURES

5.44. Radiation protection procedures should be prepared within the framework of the radiation protection programme for the facility.

5.45. Radiation protection procedures should contain instructions for implementing the radiation protection programme, such as instructions for periodic measurements (e.g. for bioassay, contamination surveys, stack effluents, source inventory). The procedures should contain acceptance criteria and should identify the person responsible for reviewing the results. A master radiation protection procedure should be prepared to ensure that all activities are completed at the required frequencies.

5.46. Radiation protection provisions for those personnel operating the reactor, conducting maintenance and periodic testing or performing experiments should be included in each of the relevant procedures and, if necessary, in the work permits for their implementation.

5.47. In developing radiation protection procedures, compliance with the OLCs for reactor operation and with regulatory body requirements should be verified.

PROCEDURES FOR THE AUTHORIZATION OF OPERATION, MAINTENANCE AND UTILIZATION

5.48. Procedures for the authorization of operation, maintenance and utilization (conduct of irradiations or performance of experiments) that could affect the safety of the reactor should be prepared, to define the conditions, the levels of responsibility and the means of authorization.

5.49. A procedure for dealing with proposals for experiments should be established. The procedure should require that the proposals include:

- (a) A description of the purpose and intended conduct of the experiment;
- (b) The means of integrating the experimental device with the reactor systems;
- (c) The selection and justification of the criteria employed in the design of the experimental device;
- (d) A safety assessment of the experimental device, including the experiment itself, and of its effects on the safety of the reactor and personnel;
- (e) The requirements for the production and validation of any special documentation for operation and maintenance;
- (f) The requirements for any special training of operating and maintenance personnel and experimenters;
- (g) The requirements for commissioning and functional testing;

- (h) Special requirements for transport, if necessary;
- (i) A decommissioning plan for the experimental device;
- (j) The quality assurance programme used;
- (k) The suggested means of disposal of the radioactive waste generated in the experiments and of the experimental devices after their final use;
- (l) Considerations relating to alarms and interlocks;
- (m) Procedures to ensure adequate means of communication between operators and experimenters.

5.50. Procedures should be prepared for the authorization of irradiations and isotope production. The procedures should include information concerning the means by which new types of irradiation (e.g. irradiation of new materials or of greater quantities of the usual materials) should be authorized.

PROCEDURES FOR OPERATOR RESPONSE TO ANTICIPATED OPERATIONAL OCCURRENCES

5.51. Procedures for guiding the response of the operator to anticipated operational occurrences, design basis accidents and, to the extent feasible, beyond design basis accident conditions should be prepared and should be periodically exercised. To improve their execution, the procedures should be reviewed and modified on the basis of operational experience and the performance of the exercises.

5.52. The procedures mentioned in para. 5.49 should be included with the operational procedures for the reactor. On the basis of the safety analysis report, the procedures should contain the duties of the operating organization for all anticipated operational occurrences. The instructions for all operating procedures should be clear and brief, and particularly so for those dealing with anticipated operational occurrences.

EMERGENCY PROCEDURES

5.53. Emergency procedures should be prepared, and they should be a component of the emergency plan. Their development should be based on the evaluation and analysis of all aspects of possible emergencies. The procedures should specify the methods and duties of intervention staff and the emergency actions that are necessary to mitigate the possible consequences of the

emergency. They should refer to the emergency facilities and emergency equipment.

5.54. Emergency drills should be periodically conducted. On the basis of experience gained from the performance of the drills, the procedures, including the list of organizations and individuals to be called or informed in the event of an emergency, should be reviewed at specified periods and amended, if necessary, to improve their execution and to ensure that the lessons learned are taken into account.

5.55. In preparing emergency procedures in which the services of off-site organizations such as hospitals, police forces, fire departments and ambulance services are utilized, agreements should be obtained from the organizations concerned. Formal letters of agreement and lists of contact points should be maintained and periodically updated. In addition, emergency procedures including emergency actions to be taken by off-site organizations should include clear and detailed instructions that have been agreed with the off-site organizations.

5.56. Experts should be consulted in preparing emergency procedures (e.g. specialized personnel from fire departments, hospital emergency units, ambulance crews, etc.) [7, 8].

PHYSICAL PROTECTION PROCEDURES

5.57. Physical protection procedures should be prepared and should be a component of a physical protection plan. The plan should be approved by the competent authorities as specified in the legal framework of the State. The plan and procedures should be kept confidential and should be revealed on a need to know basis only.

5.58. Physical protection procedures should be developed on the basis of the evaluation and analysis of all aspects of security. The instructions for physical protection procedures should be brief but should give sufficient details of the essential steps for coping with security matters.

5.59. If the services of off-site organizations such as police forces and army units are required by physical protection procedures, formal letters of agreement should be obtained and these should be periodically updated.

5.60. Experts should be consulted in preparing security procedures (e.g. security specialists, specialized personnel from the police and army).

PROCEDURES FOR THE HANDLING OF RADIOACTIVE WASTE AND CONTROL OF RADIOACTIVE RELEASES

5.61. Requirements covering the procedures for handling radioactive waste and for the monitoring and control of radioactive releases are established in Ref. [9]. These procedures may be included as part of the radiation protection procedures or may form an individual group within the operating procedures.

5.62. At some research reactor facilities, the operating personnel and radiation protection staff handle, collect, process, account for, store and dispose of radioactive waste.¹⁸ In this case the procedures for radioactive waste are usually considered to be radiation protection procedures. Further guidance on the predisposal management of radioactive waste is provided in other Safety Guides [10, 11].

5.63. Procedures for controlling radioactive discharges play an important part in environmental protection and also in gaining public acceptance of the facility. The sensitivity of this matter should be taken into account in the development of these procedures. Further guidance is provided in Ref. [12].

PROCEDURES FOR EXTENDED SHUTDOWN

5.64. During the extended shutdown period, modifications of the procedures for preventive maintenance and periodic testing may be introduced. Relief from the procedures is possible in those activities that are a function of operating time or energy generation.

5.65. Special procedures should be developed for those inspections that could indicate degradation of mechanical and electrical systems and components.

5.66. Procedures for extended shutdown should be derived from the procedures for normal operation. The effects of giving exemptions for activities

 $^{^{18}\ {\}rm Further}$ guidance is provided in the IAEA safety standards relating to radioactive waste.

to be conducted during extended shutdown should be carefully investigated, as some activities could cause significant degradation in the systems, which could prevent the recommissioning of the reactor.

5.67. Surveillance procedures to be used during extended shutdown periods for the reactor should be derived from the surveillance procedures for an operating reactor, with exemptions given for certain activities or strengthening of the procedures for others.

5.68. If the operating procedures are modified for the extended shutdown state of the facility, the original version of the operating procedures should be kept for possible future operation of the reactor.

5.69. Procedures should be developed for disconnecting, dismantling and preserving the systems that are to be taken out of operation or temporarily dismantled.

PROCEDURES FOR UTILIZATION AND FOR MODIFICATION OF THE REACTOR

5.70. Procedures should be prepared for controlling utilization or modification of the reactor¹⁹ having safety significance, to ensure that the design, manufacture, installation, conduct and testing of experiments and modifications are properly executed.

5.71. The utilization and modification of the reactor should be performed in accordance with approved procedures that contain provisions for ensuring that all OLCs are met.

5.72. The following considerations should be addressed in the modification procedure:

- (a) Description and drawings of the proposed modification;
- (b) Justification of the need for the modification;
- (c) Design requirements and criteria;
- (d) Safety assessment supporting the modification, including influences on other systems;

¹⁹ For further details, see Ref. [3].

- (e) Manufacturing processes;
- (f) Installation processes;
- (g) The commissioning process;
- (h) Review and modification of existing operating procedures and the need for new procedures;
- (i) Updating of documentation (drawings, training materials, etc.);
- (j) Training requirements for reactor operators and support staff (including requalification and relicensing, if necessary);
- (k) Quality assurance requirements;
- (1) Requirements for the optimization of radiation protection;
- (m) Requirements for physical protection;
- (n) Requirements for radioactive waste disposal.

5.73. A procedure should be prepared for the categorization of experiments and modifications on the basis of their safety significance. Generally, three categories for experiments and modifications are utilized: those that fall outside the approved OLCs (major safety significance); those that are within the OLCs (moderate safety significance); and those having no safety significance. An extensive discussion of categorization, safety assessment and approval routes for experiments and modifications is provided in section 3 of Ref. [3].

5.74. In the preparation of procedures for utilization or modification, particular attention should be paid to the consequences for safety systems and for reactor operation. The safety level of the reactor during or as a result of the modification or utilization should not be reduced below the limits set in the OLCs.

5.75. The procedures may include a system of work permits for use for modification or utilization and when performing pre-operational testing, including appropriate checks before and after the conduct of the work in accordance with the quality assurance programme. This is to ensure that all work is conducted with the knowledge and permission of the person in operational control of the reactor, and to ensure the protection and safety of the personnel doing the work as well as the safety of the reactor.

5.76. Procedures for modification and utilization should specify any changes to the normal reactor operating configuration (e.g. valve line-ups) and should include provisions for restoration of the normal configuration.

5.77. Procedures for modification and utilization should include assessment of the results of the modification or utilization by properly qualified persons to verify compliance with the approved design and with the OLCs. The procedures should require comparisons, where appropriate, with the premodification conditions to determine possible failures and to permit timely corrective action.

5.78. Modification procedures should ensure that the resumption of normal operation will be permitted only after the person who is responsible for coordinating the modification work has approved the results, including the updating of documentation.

5.79. Special procedures should be prepared to control modification work performed by contractors. These procedures should include requirements for contractors' qualifications and training and for the coordination of work, as well as provisions for surveillance of the contractors. The procedures should list activities of the contractor that are prohibited (e.g. activities causing the generation of dust, use of the reactor power supply).

ADMINISTRATIVE PROCEDURES

5.80. Administrative procedures should be developed for all operations that are of an administrative nature and that may have an effect on the safety of the reactor (e.g. personnel training and retraining, transport of radioactive material, fuel management and control of visitors).

5.81. Administrative procedures that are consistent with the quality assurance programme should be developed for the generation, collection, and retention and archiving of records and reports.

5.82. Administrative procedures may be used as implementing procedures for the quality assurance programme.

5.83. Administrative procedures should be based on national standards or national regulatory requirements. Where appropriate, as for the transport of radioactive material, the administrative procedures should be in accordance with international standards, agreements or practices.

6. TRAINING OF PERSONNEL IN THE USE OF PROCEDURES

6.1. To ensure that procedures will be properly executed, operating personnel and others should be knowledgeable in the current procedures relating to the tasks for which they are responsible.

6.2. Before a procedure is used, training of operating personnel and others who are intended to implement the procedure should be conducted. Training may take the form of oral or written instructions, demonstrations, drills, training classes or comprehensive training courses and, where applicable, the use of mock-ups.

6.3. Retraining in the use of procedures should be included in the retraining programme for operating personnel and others, and the frequency of the retraining programme should be specified. Retraining should be conducted according to a plan, and special attention should be paid to emergency procedures and infrequently performed procedures. Typical examples of these are:

- Operation of the emergency core cooling systems;
- Operation of the air cleaning system for the containment or the means of confinement;
- Testing of the leaktightness of the reactor building;
- Handling of highly radioactive material under abnormal conditions;
- Fuel shipment;
- Emergency actions such as responses to a fire alarm or evacuation alarm in the reactor building, a personal injury, the release of airborne radioactive material and weather warnings;
- Use of personal protective equipment.

6.4. The operating personnel and others should be required to demonstrate their knowledge and understanding of the operating procedures that they follow in discharging their responsibilities.

6.5. Procedures should be developed for validation of the training, to verify its effectiveness and the qualifications of the staff.

7. COMPLIANCE WITH OPERATIONAL LIMITS AND CONDITIONS AND OPERATING PROCEDURES

7.1. The operating organization has the primary responsibility for ensuring compliance with the OLCs. Relevant controls should be established to discharge this responsibility.²⁰ The provision of and training in operating procedures consistent with the OLCs is a major contribution to ensuring compliance with the OLCs. Some OLCs may be directly stated in procedures or other documents, and if so this should be clearly indicated in the implementation document.

7.2. To ensure compliance with OLCs, all persons who are responsible for such compliance should always have available a copy of the OLCs currently in force and should be adequately trained in their application. If possible, operational limits should be legibly indicated on instruments and displays to facilitate compliance. Similarly, the current operating procedures should be immediately available to the control room personnel and to others who need to use them or refer to them.

7.3. If an OLC cannot be met or a procedure cannot be followed, this should be reported and the causes should be analysed. This may lead to the modification of an OLC or procedure in accordance with established procedures that allow for changes to be made in a controlled manner and approved as required by the regulatory body. Results of routine or commissioning tests also necessitate analyses and consideration of the need for modifications to the OLCs and/or the operating procedures.

7.4. Records of operation of the research reactor and demonstrations of compliance with the OLCs and operating procedures should be kept and should be stored in accordance with the quality assurance requirements. Reports of non-compliance should be investigated to ensure that corrective action is taken and to help prevent such non-compliance in the future.

²⁰ Further guidance is provided in Ref. [6].

8. QUALITY ASSURANCE

GENERAL

8.1. A quality assurance²¹ programme based on specific quality standards for the operation and utilization of a research reactor should be established, managed, applied and evaluated for ensuring safety.²² The senior management should establish, implement and maintain the quality assurance programme, which should cover all processes of the organization, including the operational processes, the interfaces at different operational phases, and the development and implementation of operating procedures. The extent of the quality assurance programme that is required for the operating organization of a particular research reactor will depend on the safety significance of the activities and processes relating to the reactor and on the legal requirements established by the regulatory body. For further guidance on the format and content of the quality assurance programme, see Ref. [13].

8.2. The quality assurance programme should be reviewed and approved at the appropriate levels of management within the operating organization and, where necessary, by the regulatory body.

8.3. The quality assurance programme should provide for independent verification that a component or system is returned to its operational state (including verification that it is not bypassed or disabled) after an inspection, operability check, calibration or any other associated maintenance activity.

8.4. The management should ensure that an approved quality assurance programme is documented and distributed to staff for implementation of its requirements and should ensure its effective implementation.

²¹ The IAEA has revised the requirements and guidance in the subject area of quality assurance for new safety standards on management systems for the safety of nuclear facilities and activities involving the use of ionizing radiation [13, 14]. Aspects of managing a nuclear facility, including the safety, health, environmental and quality requirements, are to be integrated into one coherent management system.

²² See Ref. [1], paras 4.5–4.9.

SPECIFIC QUALITY ASSURANCE ARRANGEMENTS FOR OPERATIONAL LIMITS AND CONDITIONS

8.5. To provide assurance that the selection, development, implementation and modification of the OLCs have been correctly conceived, the overall quality assurance programme for operation of the reactor should specify the system for controlling the establishment of the OLCs and the management responsibilities for this process. This process should also be applied to the development of any additional operating rules and other technical requirements referred to in the OLCs. This quality assurance programme refers to other documents that define the processes in detail.

8.6. The items to be audited and the maximum interval between audits should be stated and may include, for example:

- Facility operations for conformance with the OLCs and the facility licence;
- The response by the operating organization to violations of the OLCs;
- The implementation of licensing procedures;
- Events that should be reported to the regulatory body;
- The facility emergency plan and procedures;
- The training programme for reactor operators;
- The updating of documents.

For requirements and guidance on auditing, see Refs [13, 14], and also Safety Guide No. 50-C/SG-Q: Q5.²³

QUALITY ASSURANCE FOR OPERATING PROCEDURES

8.7. The quality assurance programme should specify the system for the development, implementation and modification of operating procedures to ensure that they have been correctly conceived and implemented.

For requirements and guidance, see Refs [13, 14] and Safety Guides No 50-C/SG-Q: Q3, Q13 (see footnote 23).

²³ INTERNATIONAL ATOMIC ENERGY AGENCY, Quality Assurance for Safety in Nuclear Power Plants and other Nuclear Installations: Code and Safety Guides Q1–Q14, Safety Series No. 50-C/SG-Q, IAEA, Vienna (1996).

8.8. The quality assurance programme should specify the means of detecting and correcting non-conformances in the operating procedures. This may be done by means of periodic review and auditing by the operating organization.

Appendix I

FACTORS TO BE CONSIDERED IN ESTABLISHING LIMITING CONDITIONS FOR SAFE OPERATION

I.1. The following list of operational parameters and equipment should be considered in establishing limiting conditions for safe operation of research reactors. These limiting conditions for safe operation may be operational constraints or administrative limitations imposed on each of the listed items. The user should look at the entire list and select the appropriate items in accordance with the type of reactor and the conditions of operation. The grouping of items by systems or activities of a common nature has been made only for convenience. However, the presentation of limiting conditions for safe operation by groups in the OLCs document provides for a logical arrangement and gives clarity to the document.

I.2. FUEL AND FUEL ELEMENTS AND ASSEMBLIES

- (a) Uranium enrichment;
- (b) Uranium content;
- (c) Materials used;
- (d) Geometry;
- (e) Burnup limits;
- (f) Fuel failure criteria (e.g. maximum allowed activity of the cooling water);
- (g) Inspection and testing of fresh fuel and in-service elements and assemblies.

I.3. FUEL HANDLING AND STORAGE OF FRESH AND SPENT FUEL

- (a) Storage of fresh fuel;
- (b) Storage of spent fuel;
- (c) Storage of failed fuel;
- (d) Capability to unload and store core components;
- (e) Requirements for fuel movements (e.g. staffing, tools, measurements);
- (f) Requirements for preparation of fuel for off-site shipment.

I.4. REACTOR CORE CONFIGURATION

- (a) Permissible internal or peripheral cavities;
- (b) Maximum and minimum number of fuel elements;
- (c) Conditions of reflection (e.g. type of reflector and configuration);
- (d) Number of control elements, including fuel followers;
- (e) Mixed cores (e.g. cores containing fuel of different enrichments);
- (f) Permissible configurations;
- (g) Requirements for determining new configurations;
- (h) Reactor power;
- (i) Average and peak fuel element power;
- (j) Maximum allowed fuel temperature and cladding temperature;
- (k) Departure from nucleate boiling ratio or flow instability.

I.5. REACTIVITY AND REACTIVITY CONTROL SYSTEMS

- (a) Maximum excess reactivity;
- (b) Minimum shutdown margin during operation and during fuel movement;
- (c) Reactivity worth of the reactivity control mechanisms (e.g. regulating, shim, safety, pulse rods or blades);
- (d) Reactivity addition rates by means of reactivity control mechanisms, experiments and fuel elements;
- (e) Total reactivity worth of all experiments;
- (f) Maximum reactivity worth of specific types of experiment (e.g. experiments fixed or not fixed to the reactor structure);
- (g) Reactivity worth of backup shutdown system (if any);
- (h) Reactivity balance (e.g. pattern of withdrawal levels of the reactivity control mechanisms, fuel burnup distribution in the core);
- (i) Type and number of control rods (including material, configuration).

I.6. PROTECTION SYSTEMS AND REACTOR SHUTDOWN SYSTEMS

- (a) Type and minimum number of neutronic measuring equipment items necessary to scram the reactor in each mode of operation;
- (b) Type and minimum number of other measuring equipment items (temperature, flow, radiation level, etc.) necessary to scram the reactor;
- (c) Alarm and scram limits for the aforementioned equipment;
- (d) Interlocks and trips;
- (e) Bypassing channels;
- (f) Other safety instrumentation;
- (g) Reactor shutdown delay time (e.g. rod drop time).

I.7. REACTOR STARTUP AND OPERATION

- (a) Minimum operability requirements of structures, systems and components;
- (b) Completion and review of checklists;
- (c) Visual inspections of reactor core, beam tube shutters and shielding;
- (d) Additional conditions for startup following a trip.

I.8. COOLANT SYSTEMS AND CONNECTED SYSTEMS

- (a) Coolant chemistry (content of solids and dissolved gases, pH and conductivity);
- (b) Temperature, pressure (in lines, across filters, etc.) and flow;
- (c) System configuration for different modes of operation (e.g. how many and which pumps should be operable, which main valves should be open or closed, etc.);
- (d) Changeover conditions to and from the natural convection mode of cooling, if applicable;
- (e) Coolant or moderator level;
- (f) Emergency core cooling;
- (g) Leak detection and loss of coolant alarm limits;
- (h) Radionuclide content in the coolant;
- (i) Content of fission products in the coolant;
- (j) Coolant availability;
- (k) Ultimate heat sink;
- (1) Moderator chemistry (e.g. required properties and characteristics).

I.9. CONTAINMENT SYSTEMS OR MEANS OF CONFINEMENT, INCLUDING VENTILATION

- (a) Temperature, humidity and air flow in different areas of the reactor;
- (b) Pressure drop across filters;
- (c) Containment pressure relative to the atmosphere (normal and under emergency conditions);
- (d) Isolation of the containment or the means of confinement and starting of emergency ventilation;
- (e) Operations that require containment or confinement;
- (f) Configuration and minimum equipment for ventilation;
- (g) Leak rate from the containment or the means of confinement;
- (h) Hazardous materials inside the containment or the means of confinement;
- (i) Efficiencies of filters and iodine traps.

I.10. OPERATIONAL RADIATION PROTECTION

- (a) Type (gaseous, particulate, gamma, neutron, etc.) and location of radiation monitoring instruments;
- (b) Alarm setting for monitoring instruments for radiation (including monitoring instruments for initiating scrams, if any);
- (c) Limits on the concentration of radionuclides or other limits on the liquid or gaseous effluents that may be released in a given time period, such as maximum annual releases (site limits may apply where more than one facility is located at the same site);
- (d) Dose control values for operation, such as annual dose limits;
- (e) Operating limits for surface contamination;
- (f) Dose targets (individual and collective);
- (g) Criteria for respiratory protection and special protective clothing;
- (h) Criteria for bioassay or whole body counting;
- (i) Storage capacity for liquid and solid waste.

I.11. INSTRUMENTATION AND CONTROL SYSTEMS

- (a) Type and minimum number of items of measuring equipment associated with safety systems;
- (b) Startup instrumentation;
- (c) Display monitors;
- (d) Data acquisition systems;
- (e) Requirements for the calibration of instrumentation and its periodic control, including updating of the related documentation.

I.12. EXPERIMENTAL DEVICES

- (a) Suitability of materials for use in the ambient conditions, for encapsulation of irradiation samples, with fissile materials, etc.;
- (b) Explosive materials;
- (c) Interlocking requirements for experiments.

I.13. ELECTRIC POWER SUPPLY SYSTEMS

- (a) Emergency power supplies for all operational states (e.g. configuration of distributors and list of equipment connected to a distributor, startup and operation of diesel generators, batteries for the uninterruptible power supply system, etc.);
- (b) Testing of emergency power supplies.

I.14. AUXILIARY SYSTEMS AND EQUIPMENT

- (a) Fire protection systems;
- (b) Communication systems;
- (c) Cranes (e.g. limitation of manipulation and loading);
- (d) Emergency lighting systems.

I.15. OTHER LIMITATIONS

- (a) Other design features;
- (b) Site features;
- (c) Administrative controls.

Appendix II

INDICATIVE LIST OF OPERATING PROCEDURES

II.1. COMMISSIONING PROCEDURES

- (a) Cleaning procedures for the cooling and ventilation systems;
- (b) Commissioning and acceptance tests for mechanical, electrical and instrumentation systems and components;
- (c) Tests prior to fuel loading;
- (d) Fuel handling;
- (e) Fuel loading, initial criticality tests and low power tests;
- (f) Control rod calibration;
- (g) Calibration of reactor safety channels and other neutron detection channels;
- (h) Determination of shutdown margin and core excess reactivity;
- (i) Power ascension tests and power tests;
- (j) Determination of thermal power level;
- (k) Reactor emergency procedures;
- (l) Radiation protection procedures;
- (m) Procedures for removal of foreign material;
- (n) Verification of biological shielding;
- (o) Handling of non-conformances.

II.2. OPERATIONAL PROCEDURES

- (a) Reactor startup, operation, power level changes and shutdown;
- (b) Determination of core reactivity to meet the OLCs (e.g. shutdown margin and excess reactivity, set point calculations, etc.);
- (c) Determination of thermal power level;
- (d) Routine loading, unloading, handling and movement of fuel elements, fuel assemblies and other core components and reflector components;
- (e) Special fuel handling (e.g. handling of failed fuel, preparation for and shipment of spent fuel);
- (f) Reactivity determination, loading, unloading, irradiation, handling and safety evaluation for experimental devices;
- (g) Performance of routine checks on reactor operation, the status of systems and the condition of the facility;
- (h) Operation of the mechanical and electrical support systems and equipment of the reactor;

- (i) Shift turnover entries and entries in the reactor logbooks;
- (j) Acceptance testing of new fuel elements;
- (k) Ion exchange replacement or regeneration;
- (l) Operation of hoisting devices.

II.3. MAINTENANCE PROCEDURES

- (a) Equipment replacement and repair;
- (b) Preventive maintenance of reactor equipment and reactor support system equipment;
- (c) Repair or replacement of reactivity control mechanisms;
- (d) Cleaning of the heat exchanger and tube plugging;
- (e) Replacement of seals;
- (f) Replacement of filters;
- (g) Routine maintenance on the overhead crane;
- (h) Routine maintenance on the auxiliary power supply;
- (i) In-service inspection;
- (j) Inventories of spare parts.

II.4. INSPECTION, CALIBRATION AND PERIODIC TESTING PROCEDURES

- (a) Inspection and dimensional checking of fuel and preparation of the fuel inventory;
- (b) Inspection and calibration of the reactivity control mechanism;
- (c) Release and drop time measurements for safety rods and control rods;
- (d) Calibration of reactor measuring channels and protection channels;
- (e) Calibration and testing of fixed and portable radiation monitors, airborne radiation monitors and personal monitors;
- (f) Calibration and testing of process systems (e.g. for temperature, flow, emergency power and ventilation);
- (g) Performance check for emergency core cooling systems;
- (h) Efficiency and flow rate measurements for the emergency ventilation system;
- (i) Flow rate measurements for the operational ventilation system;
- (j) In situ periodic testing of built-in filters in the ventilation system;
- (k) Measurements in the primary cooling system;
- (1) In-service inspection of the reactor vessel, pool liner and core components;
- (m) Performance checks for the secondary cooling system;
- (n) Measurement of the leaktightness of the building;

- (o) Testing of pneumatic tube systems;
- (p) Testing of the emergency power supply system;
- (q) Checking of the fire protection system.

II.5. RADIATION PROTECTION PROCEDURES

- (a) Radiation surveys and air sampling;
- (b) Control of the contamination of surfaces, personnel and equipment, including the use of decontamination facilities;
- (c) Administrative measures for controlling access to or residence time in radiation areas;
- (d) Control of occupational radiation exposure, such as procedures for monitoring external and internal exposures of the operating personnel, temporary personnel and visitors;
- (e) Issue, selection, use and maintenance of protective equipment;
- (f) Monitoring of radioactive material and the packaging and shipment of radioactive material;
- (g) Analysis of the reactor coolant;
- (h) Inventory, handling and leak testing of stored sealed radioactive sources;
- (i) Control and periodic review of operations to ensure that radiation exposures are as low as reasonably achievable.

II.6. PROCEDURES FOR THE AUTHORIZATION OF OPERATION, MAINTENANCE, IRRADIATION OR EXPERIMENTS

- (a) Authorization for operation;
- (b) Authorization for maintenance (work permit);
- (c) Authorization for modification;
- (d) Authorization for experiments;
- (e) Permit for irradiation;
- (f) Procedures for isotope production.

II.7. PROCEDURES FOR OPERATOR RESPONSE TO ANTICIPATED OPERATIONAL OCCURRENCES

- (a) Operator response to alarms, loss of electric power supplies, instrument failures, pipe leakage, etc.;
- (b) Operator response to failures of experiments;
- (c) Response to abnormal radioactive releases;
- (d) Response to spread of contamination.

II.8. EMERGENCY PROCEDURES

- (a) High airborne radiation levels or area radiation levels;
- (b) Fire, internal flooding;
- (c) Tornado, hurricane, typhoon, flood or other weather related emergency;
- (d) Earthquake;
- (e) Injury of personnel, with and without radioactive contamination;
- (f) Credible reactor accident (e.g. loss of primary coolant, abnormal release of radioactive material, rapid insertion of positive reactivity, significant fuel failure);
- (g) Aircraft crash, sabotage or attempted sabotage.

II.9. PHYSICAL PROTECTION PROCEDURES

- (a) Surveillance and alarm system tests for fuel storage areas;
- (b) Surveillance and alarm system tests for facility access points;
- (c) Required patrols and inspections during reactor operation and when the reactor is shut down;
- (d) Control of access to the facility (e.g. identification badges, door locks, closed circuit television monitoring systems, electronic card keys);
- (e) Preventing and/or coping with an unauthorized intruder;
- (f) Coping with an attack;
- (g) Coping with a civil disturbance.

II.10. PROCEDURES FOR THE HANDLING OF RADIOACTIVE WASTE AND CONTROL OF RADIOACTIVE RELEASES

- (a) Monitoring, handling, storage and disposal of solid radioactive waste;
- (b) Collection, monitoring, processing and disposal of liquid radioactive waste;
- (c) Control of monitoring for gaseous and particulate airborne radioactive releases.

II.11. PROCEDURES FOR EXTENDED SHUTDOWN

- (a) Disconnecting and dismantling of systems that have to be taken out of service;
- (b) Protecting systems or components against deterioration;
- (c) Preventing undesired use of systems or components that have been taken out of service (e.g. electric power supply, isolation valves).

II.12. DECOMMISSIONING PROCEDURES

- (a) Handling of spent fuel;
- (b) Dismantling and handling of activated and radioactively contaminated components, and handling of radioactive waste;
- (c) Radiation protection during and after decommissioning.

II.13. PROCEDURES FOR UTILIZATION AND FOR MODIFICATION OF THE REACTOR

- (a) Proposals for experiments or modifications;
- (b) Determination of the safety significance of an experiment or a modification, for purposes of review and approval;
- (c) Review and approval of a new or modified experiment or modification;
- (d) Conduct of an experiment, including commissioning and decommissioning;
- (e) Conduct of a modification, including commissioning of modified systems and updating of the facility documentation.

II.14. ADMINISTRATIVE PROCEDURES

- (a) Control and accountability of nuclear material;
- (b) Reporting of the fuel inventory;
- (c) Shipment of spent fuel;
- (d) Testing and certification of packages used for the shipment of radioactive material;
- (e) Qualification, training and retraining of personnel;
- (f) Generation, collection and retention of records;
- (g) Isolation and tagging of equipment;
- (h) Instructions to personnel concerning possible health effects of radiation exposure, legal requirements, etc.;
- (i) Instructions for internal communications;
- (j) Requirement for on-call personnel;
- (k) Quality assurance;
- (1) Purchase of safety related components.

REFERENCES

- [1] INTERNATIONAL ATOMIC ENERGY AGENCY, Safety of Research Reactors, IAEA Safety Standards Series No. NS-R-4, IAEA, Vienna (2005).
- [2] INTERNATIONAL ATOMIC ENERGY AGENCY, Safety Assessment of Research Reactors and Preparation of the Safety Analysis Report, Safety Series No. 35-G1, IAEA, Vienna (1994).
- [3] INTERNATIONAL ATOMIC ENERGY AGENCY, Safety in the Utilization and Modification of Research Reactors, Safety Series No. 35-G2, IAEA, Vienna (1994).
- [4] INTERNATIONAL ATOMIC ENERGY AGENCY, Regulatory Inspection of Nuclear Facilities and Enforcement by the Regulatory Body, IAEA Safety Standards Series No. GS-G-1.3, IAEA, Vienna (2002).
- [5] INTERNATIONAL ATOMIC ENERGY AGENCY, IAEA Safety Glossary: Terminology Used in Nuclear Safety and Radiation Protection, 2007 Edition, IAEA, Vienna (2007).
- [6] INTERNATIONAL ATOMIC ENERGY AGENCY, Maintenance, Periodic Testing and Inspection of Research Reactors, IAEA Safety Standards Series No. NS-G-4.2, IAEA, Vienna (2006).
- FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED [7] NATIONS, **INTERNATIONAL** ATOMIC ENERGY AGENCY, INTERNATIONAL LABOUR ORGANIZATION, OECD NUCLEAR ENERGY AGENCY, PAN AMERICAN HEALTH ORGANIZATION, UNITED NATIONS OFFICE FOR THE COORDINATION OF HUMANITARIAN AFFAIRS, WORLD HEALTH ORGANIZATION, Preparedness and Response for a Nuclear or Radiological Emergency, IAEA Safety Standards Series No. GS-R-2, IAEA, Vienna (2002).
- FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED [8] NATIONS, **INTERNATIONAL** ATOMIC ENERGY AGENCY, INTERNATIONAL LABOUR OFFICE, PAN AMERICAN HEALTH ORGANIZATION, UNITED OFFICE NATIONS FOR THE COORDINATION OF HUMANITARIAN AFFAIRS, WORLD HEALTH ORGANIZATION, Arrangements for Preparedness for a Nuclear or Radiological Emergency, IAEA Safety Standards Series No. GS-G-2.1, IAEA, Vienna (2007).
- [9] INTERNATIONAL ATOMIC ENERGY AGENCY, Predisposal Management of Radioactive Waste, Including Decommissioning, IAEA Safety Standards Series No. WS-R-2, IAEA, Vienna (2000).
- [10] INTERNATIONAL ATOMIC ENERGY AGENCY, Predisposal Management of Low and Intermediate Level Radioactive Waste, IAEA Safety Standards Series No. WS-G-2.5, IAEA, Vienna (2003).
- [11] INTERNATIONAL ATOMIC ENERGY AGENCY, Predisposal Management of High Level Radioactive Waste, IAEA Safety Standards Series No. WS-G-2.6, IAEA, Vienna (2003).

- [12] INTERNATIONAL ATOMIC ENERGY AGENCY, Regulatory Control of Radioactive Discharges to the Environment, IAEA Safety Standards Series No. WS-G-2.3, IAEA, Vienna (2000).
- [13] INTERNATIONAL ATOMIC ENERGY AGENCY, The Management System for Facilities and Activities, IAEA Safety Standards Series No. GS-R-3, IAEA, Vienna (2006).
- [14] INTERNATIONAL ATOMIC ENERGY AGENCY, Application of the Management System for Facilities and Activities, IAEA Safety Standards Series No. GS-G-3.1, IAEA, Vienna (2006).

CONTRIBUTORS TO DRAFTING AND REVIEW

Abou Yehia, H.	Institut de radioprotection et de sûreté nucléaire, France
Adams, A.	Nuclear Regulatory Commission, United States of America
Adamski, L.	Institute of Atomic Energy, Poland
Addae, A.K.	Ghana Atomic Energy Commission, Ghana
Alcalá-Ruiz, F.	International Atomic Energy Agency
Allen, M.	Nuclear Safety Bureau, Australia
Arkhangelsky, N.	MINATOM, Russian Federation
Axmann, A.	Hahn–Meitner Institut für Kernforschung Berlin GmbH, Germany
Böck, H.	Atominstitut der Österreichischen Universitäten, Austria
Boogaard, J.P.	ECN Petten, Netherlands
Burn, R.	United States of America
Byszewski, W.	National Inspectorate for Radiation and Nuclear Safety, Poland
Cardeira, F.	Science, Technology, and Higher Education Ministry, Nuclear and Technological Institute (ITN), Portugal
Ciocanescu, M.	Pitești Nuclear Centre, Romania
Corre, J.	Commissariat à l'énergie atomique, France
De Lorenzo, N.	Centro Atómico Bariloche, Comisión Nacional de Energía Atómica, Argentina
De Oliveira Neto, J.M.	IPEN-CNEN/SPO, Brazil

DiMeglio, A.F.	International Atomic Energy Agency
Dimic, V.	Jozef Stefan Institute, Slovenia
Gazit, M.	Israel Atomic Energy Commission, Israel
Griesmayer, E.	Atominstitut der Österreichischen Universitäten, Austria
Gurewitz, Y.	Nuclear Research Centre Negev, IAEC, Israel
Haack, K.	Risø National Laboratory, Denmark
Hammer, J.	Paul Scherrer Institut, Switzerland
Hargitai, T.	International Atomic Energy Agency
Hirshfeld, H.	International Atomic Energy Agency
Klinov, A.V.	Research Institute of Atomic Reactors, Russian Federation
Koonen, E.	Centre d'Études Nucléaires, SCK-CEN, Belgium
	Contro a Diados Francis, S orr ODF, Dergram
Krull, W.	GKSS Forschungszentrum Geesthacht GmbH, Germany
	GKSS Forschungszentrum Geesthacht GmbH,
Krull, W.	GKSS Forschungszentrum Geesthacht GmbH, Germany
Krull, W. Krzysztoszek, G.	GKSS Forschungszentrum Geesthacht GmbH, Germany Institute of Atomic Energy, Poland
Krull, W. Krzysztoszek, G. Litai, D.	GKSS Forschungszentrum Geesthacht GmbH, Germany Institute of Atomic Energy, Poland International Atomic Energy Agency
Krull, W. Krzysztoszek, G. Litai, D. Manzella, P.	GKSS Forschungszentrum Geesthacht GmbH, Germany Institute of Atomic Energy, Poland International Atomic Energy Agency ENEA-DCSNPS, Italy
Krull, W. Krzysztoszek, G. Litai, D. Manzella, P. McAuley, S.	GKSS Forschungszentrum Geesthacht GmbH, Germany Institute of Atomic Energy, Poland International Atomic Energy Agency ENEA-DCSNPS, Italy Atomic Energy of Canada Limited, Canada
Krull, W. Krzysztoszek, G. Litai, D. Manzella, P. McAuley, S. Nedelik, A.	GKSS Forschungszentrum Geesthacht GmbH, Germany Institute of Atomic Energy, Poland International Atomic Energy Agency ENEA-DCSNPS, Italy Atomic Energy of Canada Limited, Canada Forschungszentrum Seibersdorf, Austria

Ohnishi, N.	Tokai Research Establishment, Japan
Øyan, R.	OECD Halden Reactor Project, Norway
Pina Santos, C.	Nuclear and Energy Research Institute, Brazil
Saad, E.A.	Atomic Energy Authority, Egypt
Sági, L.	KFKI Atomic Energy Research Institute, Hungary
Salmenhaara, S.	VTT Chemical Technology, Finland
Sharma, S.K.	Bhabha Atomic Research Centre, India
Shokr, A.M.	International Atomic Energy Agency
Soucek, V.	State Office for Nuclear Safety, Czech Republic
Sumita, K.	Osaka University, Japan
Sykes, D.H.	Atomic Energy Control Board, Canada
Thompson, W.	Nuclear Installation Inspectorate, United Kingdom
Voth, M.	Nuclear Regulatory Commission, United States of America

BODIES FOR THE ENDORSEMENT OF IAEA SAFETY STANDARDS

An asterisk denotes a corresponding member. Corresponding members receive drafts for comment and other documentation but they do not generally participate in meetings.

Commission on Safety Standards

Argentina: Oliveira, A.; Australia: Loy, J.; Brazil: Souza de Assis, A.; Canada: Pereira, J.K.; China: Li, G.; Czech Republic: Drábová, D.; Denmark: Ulbak, K.; Egypt: Abdel-Hamid, S.B.; France: Lacoste, A.-C. (Chairperson); Germany: Majer, D.; India: Sharma, S.K.; Israel: Levanon, I.; Japan: Abe, K.; Korea, Republic of: Eun, Y.-S.; Pakistan: Hashmi, J.; Russian Federation: Malyshev, A.B.; South Africa: Magugumela, M.T.; Spain: Azuara, J.A.; Sweden: Holm, L.-E.; Switzerland: Schmocker, U.; United Kingdom: Weightman, M.; United States of America: Virgilio, M.; European Commission: Waeterloos, C.; IAEA: Delattre, D. (Coordinator); International Commission on Radiological Protection: Holm, L.-E.; OECD Nuclear Energy Agency: Tanaka, T.

Nuclear Safety Standards Committee

Argentina: Sajaroff, P.; Australia: MacNab, D.; Austria: Sholly, S.; Belgium: Govaerts, P.; Brazil: de Queiroz Bogado Leite, S.; *Bulgaria: Gladychev, Y.; Canada: Newland, D.; China: Wang, J.; Croatia: Valčić, I.; *Cyprus: Demetriades, P.; Czech Republic: Böhm, K.; Egypt: Aly, A.I.M.; Finland: Reiman, L. (Chairperson); France: Saint Raymond, P.; Germany: Herttrich, M.; *Greece: Camarinopoulos, L.; Hungary: Vöröss, L.; India: Kushwaha, H.S.; Iran, Islamic Republic of: Alidousti, A.; *Iraq: Khalil Al-Kamil, A.-M.; Ireland: Hone, C.; Israel: Hirshfeld, H.; Italy: Bava, G.; Japan: Nakamura, K.; Korea, Republic of: Kim, H.-K.; Lithuania: Demčenko, M.; Mexico: González Mercado, V.; Netherlands: Jansen, R.; Pakistan: Habib, M.A.; Paraguay: Troche Figueredo, G.D.; *Peru: Ramírez Quijada, R.; Portugal: Marques, J.J.G.; Romania: Biro, L.; Russian Federation: Shvetsov, Y.E.; Slovakia: Uhrik, P.; Slovenia: Levstek, M.F.; South Africa: Bester, P.J.; Spain: Zarzuela, J.; Sweden: Hallman, A.; Switzerland: Aeberli, W.; *Thailand: Tanipanichskul, P.; Turkey: Bezdegumeli, U.; Ukraine: Bezsalyi, V.; United Kingdom: Vaughan, G.J.; United States of America: Mayfield, M.E.; European Commission: Vigne, S.; IAEA: Feige, G. (Coordinator); International Organization for Standardization:

Nigon, J.L.; *OECD Nuclear Energy Agency*: Reig, J.; *World Nuclear Association: Saint-Pierre, S.

Radiation Safety Standards Committee

Argentina: Rojkind, R.H.A.; Australia: Melbourne, A.; *Belarus: Rydlevski, L.; Belgium: Smeesters, P.; Brazil: Rodriguez Rochedo, E.R.; *Bulgaria: Katzarska, L.; Canada: Clement, C.; China: Yang, H.; Costa Rica: Pacheco Jimenez, R.; Cuba: Betancourt Hernandez, L.; *Cyprus: Demetriades, P.; Czech Republic: Petrova, K.; Denmark: Ohlenschlager, M.; *Egypt: Hassib, G.M; Finland: Markkanen, M.; France: Godet, J.; Germany: Landfermann, H.; *Greece: Kamenopoulou, V.; Hungary: Koblinger, L.; Iceland: Magnusson, S. (Chairperson); India: Sharma, D.N.; Indonesia: Akhadi, M.; Iran, Islamic Republic of: Rastkhah, N.; *Iraq: Khalil Al-Kamil, A.-M.; Ireland: Colgan, T.; Israel: Laichter, Y.; Italy: Bologna, L.; Japan: Yoda, N.; Korea, Republic of: Lee, B.; Latvia: Salmins, A.; Malaysia: Rehir, D.; Mexico: Maldonado Mercado, H.; Morocco: Tazi, S.; Netherlands: Zuur, C.; Norway: Saxebol, G.; Pakistan: Mehboob, A.E.; Paraguay: Idoyago Navarro, M.; Philippines: Valdezco, E.; Portugal: Dias de Oliviera, A.; Romania: Rodna, A.; Russian Federation: Savkin, M.; Slovakia: Jurina, V.; Slovenia: Sutej, T.; South Africa: Olivier, J.H.I.; Spain: Amor, I.; Sweden: Hofvander, P.; Switzerland: Pfeiffer, H.J.; *Thailand: Wanitsuksombut, W.; Turkey: Okyar, H.; Ukraine: Holubiev, V.; United Kingdom: Robinson, I.; United States of America: Miller, C.; European Commission: Janssens, A.; Food and Agriculture Organization of the United Nations: Byron, D.; IAEA: Boal, T. (Coordinator); International Commission on Radiological Protection: Valentin, J.; International Labour Office: Niu, S.; International Organization for Standardization: Perrin, M.; OECD Nuclear Energy Agency: Lazo, T.; Pan American Health Organization: Jimenez, P.; United Nations Scientific Committee on the Effects of Atomic Radiation: Crick, M.; World Health Organization: Carr, Z.; World Nuclear Association: Saint-Pierre, S.

Transport Safety Standards Committee

Argentina: López Vietri, J.; Australia: Sarkar, S.; Austria: Kirchnawy, F.; Belgium: Cottens, E.; Brazil: Mezrahi, A.; Bulgaria: Bakalova, A.; Canada: Faille, S.; China: Qu, Z.; Croatia: Kubelka, D.; Cuba: Quevedo Garcia, J.R.; *Cyprus: Demetriades, P.; Czech Republic: Ducháček, V.; Denmark: Breddan, K.; *Egypt: El-Shinawy, R.M.K.; Finland: Tikkinen, J.; France: Aguilar, J.; Germany: Rein, H.; *Greece: Vogiatzi, S.; Hungary: Sáfár, J.; India: Agarwal, S.P.; Iran, Islamic Republic of: Kardan, M.R.; *Iraq: Khalil

Al-Kamil, A.-M.; Ireland: Duffy, J. (Chairperson); Israel: Koch, J.; Italy: Trivelloni, S.; Japan: Amano, M.; Korea, Republic of: Kim, Y.-J.; Malaysia: Sobari, M.P.M.; Netherlands: Van Halem, H.; New Zealand: Ardouin, C.; Norway: Hornkjøl, S.; Pakistan: Rashid, M.; Paraguay: More Torres, L.E.; Philippines: Kinilitan-Parami, V.; Portugal: Buxo da Trindade, R.; Romania: Vieru, G.; Russian Federation: Ershov, V.N.; South Africa: Jutle, K.; Spain: Zamora Martin, F.; Sweden: Dahlin, G.; Switzerland: Knecht, B.; *Thailand: Wanitsuksombut, W.; *Turkey*: Ertürk, K.; *Ukraine*: Sakalo, V.; *United Kingdom*: Young, C.N.; United States of America: Brach, W.E.; Boyle, R.; European Commission: Venchiarutti, J.-C.; International Air Transport Association: Abouchaar, J.; IAEA: Wangler, M.E. (Coordinator); International Civil Aviation Organization: Rooney, K.; International Federation of Air Line Pilots' Associations: Tisdall, A.; International Maritime Organization: Rahim, I.; International Organization for Standardization: Malesys, P.; United Nations Economic Commission for Europe: Kervella, O.; Universal Postal Union: Giroux, P.; World Nuclear Transport Institute: Green, L.

Waste Safety Standards Committee

Argentina: Siraky, G.; Australia: Williams, G.; Austria: Hohenberg, J.; Belgium: Baekelandt, L.; Brazil: Heilbron, P.; *Bulgaria: Simeonov, G.; Canada: Lojk, R.; China: Fan, Z.; Croatia: Subasic, D.; Cuba: Salgado Mojena, M.; *Cyprus: Demetriades, P.; *Czech Republic: Lieteva, P.; Denmark: Nielsen, C.; *Egypt: El-Adham, K.E.A.; Finland: Ruokola, E.; France: Cailleton, R.; Hungary: Czoch, I.; India: Raj, K.; Indonesia: Yatim, S.; Iran, Islamic Republic of: Ettehadian, M.; *Iraq: Abass, H.; Israel: Dody, A.; Italy: Dionisi, M.; Japan: Ito, Y.; Korea, Republic of: Park, W.; *Latvia: Salmins, A.; Lithuania: Paulikas, V.; Mexico: Aguirre Gómez, J.; Morocco: Soufi, I.; Netherlands: Selling, H.; *Norway: Sorlie, A.; Pakistan: Rehman, R.; Paraguay: Facetti Fernandez, J.; Portugal: Flausino de Paiva, M.; Romania: Tuturici, I.; Russian Federation: Poluektov, P.P.; Slovakia: Konečný, L.; Slovenia: Mele, I.; South Africa: Pather, T. (Chairperson); Spain: Sanz, M.; Sweden: Wingefors, S.; Switzerland: Zurkinden, A.; Turkey: Özdemir, T.; Ukraine: Iievlev, S.; United Kingdom: Wilson, C.; United States of America: Camper, L.; European Commission: Hilden, W.; IAEA: Hioki, K. (Coordinator); International Organization for Standardization: Hutson, G.; OECD Nuclear Energy Agency: Riotte, H.; World Nuclear Association: Saint-Pierre, S.

Safety through international standards

"The IAEA's standards have become a key element of the global safety regime for the beneficial uses of nuclear and radiation related technologies.

"IAEA safety standards are being applied in nuclear power generation as well as in medicine, industry, agriculture, research and education to ensure the proper protection of people and the environment."

> Mohamed ElBaradei IAEA Director General

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA ISBN 978–92–0–103508–0 ISSN 1020–525X